



Article Functional Brain Network Disruptions in Parkinson's Disease: Insights from Information Theory and Machine Learning

Ömer Akgüller ^{1,2}, Mehmet Ali Balcı ^{1,*} and Gabriela Cioca ³

- ¹ Faculty of Science, Department of Mathematics, Mugla Sitki Kocman University, Muğla 48000, Turkey; oakguller@mu.edu.tr or omer.akguller@ikc.edu.tr
- ² Engineering Sciences Department, Engineering and Architecture Faculty, Izmir Katip Celebi University, Izmir 35620, Turkey
- ³ Preclinical Department, Faculty of Medicine, Lucian Blaga University of Sibiu, 550024 Sibiu, Romania; gabriela.cioca@ulbsibiu.ro
- * Correspondence: mehmetalibalci@mu.edu.tr

Abstract: Objectives: This study investigates disruptions in functional brain networks in Parkinson's Disease (PD), using advanced modeling and machine learning. Functional networks were constructed using the Nonlinear Autoregressive Distributed Lag (NARDL) model, which captures nonlinear and asymmetric dependencies between regions of interest (ROIs). Key network metrics and information-theoretic measures were extracted to classify PD patients and healthy controls (HC), using deep learning models, with explainability methods employed to identify influential features. Methods: Resting-state fMRI data from the Parkinson's Progression Markers Initiative (PPMI) dataset were used to construct NARDL-based networks. Metrics, such as Degree, Closeness, Betweenness, and Eigenvector Centrality, along with Network Entropy and Complexity, were analyzed. Convolutional Neural Networks (CNNs), Recurrent Neural Networks (RNNs), and Long Short-Term Memory (LSTM) models, classified PD and HC groups. Explainability techniques, including SHAP and LIME, identified significant features driving the classifications. Results: PD patients showed reduced Closeness (22%) and Betweenness Centrality (18%). CNN achieved 91% accuracy, with Network Entropy and Eigenvector Centrality identified as key features. Increased Network Entropy indicated heightened randomness in PD brain networks. Conclusions: NARDL-based analysis with interpretable deep learning effectively distinguishes PD from HC, offering insights into neural disruptions and potential personalized treatments for PD.

Keywords: Parkinson's disease; brain network; information theory; explainable deep learning

1. Introduction

Parkinson's Disease (PD) is a chronic, progressive neurodegenerative disorder characterized predominantly by motor symptoms, such as bradykinesia, rigidity, resting tremor, and postural instability, which result from the loss of dopaminergic neurons in the substantia nigra pars compacta of the midbrain [1,2]. While these motor manifestations are the most visible and have historically been the primary focus for diagnosis and treatment, PD is increasingly recognized as a multisystem disorder that affects various non-motor domains, including cognitive, psychiatric, autonomic, and sensory functions [3,4].

Cognitive impairments in PD range from mild cognitive impairment to Parkinson's Disease Dementia and significantly impact patients' quality of life and daily functioning [5,6]. Understanding the neural mechanisms underlying cognitive decline in PD is crucial for the early detection, prognosis, and the development of targeted therapeutic interventions [7].

Functional connectivity refers to the temporal correlations and statistical dependencies between spatially remote neurophysiological events, which provide insights into the coordinated activity of different brain regions [8]. Functional Magnetic Resonance Imaging



Citation: Akgüller, Ö.; Balcı, M.A.; Cioca, G. Functional Brain Network Disruptions in Parkinson's Disease: Insights from Information Theory and Machine Learning. *Diagnostics* **2024**, *14*, 2728. https://doi.org/10.3390/ diagnostics14232728

Academic Editor: Rahib Abiyev

Received: 26 September 2024 Revised: 18 November 2024 Accepted: 2 December 2024 Published: 4 December 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). (fMRI) has emerged as a pivotal tool for investigating functional connectivity, leveraging blood-oxygen-level-dependent (BOLD) signals to infer neural activity [9,10]. Resting-state fMRI allows for the examination of intrinsic brain networks without task-induced confounds, offering a window into the brain's functional organization in both health and disease [11,12].

Graph-theoretical approaches have been widely applied to fMRI data to construct functional brain networks, where nodes represent regions of interest (ROIs) and edges represent functional connections based on statistical associations [13,14]. These methods enable the quantification of network properties such as efficiency, modularity, and centrality, facilitating the exploration of how brain networks reorganize in PD [15,16]. Studies have reported alterations in network topology in PD patients, including decreased global efficiency, disrupted small-world properties, and altered connectivity within specific networks like the default mode network and the frontoparietal network [17–19]. These network changes are associated with both motor and cognitive symptoms, suggesting that functional connectivity alterations underpin the clinical manifestations of PD [20,21].

However, traditional graph-theoretical analyses often rely on linear correlation measures, such as Pearson's correlation, to estimate functional connectivity [8]. While these methods are valuable, they may not fully capture the complex, nonlinear, and dynamic interactions within the brain, particularly in pathological conditions like PD [22]. Neural systems exhibit both linear and nonlinear dependencies, and their interactions can be asymmetric and time-varying, especially in PD [23]. Therefore, advanced analytical techniques that can model these complexities are essential for a more accurate and comprehensive understanding of functional connectivity alterations in PD.

This study aims to bridge this gap by employing the Nonlinear Autoregressive Distributed Lag (NARDL) model to construct functional brain networks from resting-state fMRI data of PD patients and healthy controls (HC). By capturing both nonlinear and asymmetric dependencies between ROIs, the NARDL model provides a nuanced representation of neural interactions that traditional linear models may overlook. Additionally, the study integrates information-theoretic measures to quantify the informational properties of brain networks, offering deeper insights into the structural and functional disruptions caused by PD. Coupled with explainable deep learning models, this approach not only enhances the detection and classification of PD but also ensures that the findings are interpretable and clinically relevant. These advancements contribute to the development of more accurate diagnostic tools and personalized treatment strategies, ultimately improving patient outcomes and advancing our understanding of PD's impact on brain connectivity.

This study is structured as follows: Section 2 reviews the existing literature on functional connectivity in PD, highlighting gaps such as the underutilization of nonlinear modeling and explainable machine learning. Section 3 details the participant cohort, data acquisition, application of the NARDL model for constructing functional brain networks, and the extraction of connectivity and information-theoretic measures. Section 4 presents the findings from statistical analyses and deep learning models; detailed discussions interpreting these results are provided immediately afterward within the same section to contextualize and understand their implications. Finally, Section 5 summarizes the key contributions of the study, emphasizing how integrating NARDL-based modeling, information-theoretic measures, and explainable deep learning advances our understanding of functional brain network alterations in PD and can inform clinical practices.

2. Related Works

In Parkinson's disease, functional connectivity studies reveal significant insights into both motor and non-motor symptoms, including cognitive decline. Resting-state fMRI studies, for instance, have consistently shown altered connectivity within the default mode network, sensorimotor networks, and other brain regions, correlating these changes with cognitive and motor impairments in PD patients [24–26]. These findings suggest that disruptions in functional networks may serve as early indicators of disease progression, making functional connectivity a valuable biomarker for tracking neurodegeneration.

Studies using whole-brain functional connectivity analyses highlight reduced connectivity in regions associated with cognition and motor control, such as the fronto-striatal and visual networks, which often precede notable cognitive and functional deficits [27,28]. Additionally, dynamic functional connectivity, which measures the temporal fluctuations of functional connectivity, has shown that patients with advanced cognitive impairment in PD tend to experience a higher prevalence of segregated network states, limiting inter-network communication necessary for cognitive flexibility [29,30].

Research by Ref. [27] illustrated abnormal connectivity patterns in PD patients, identifying compensatory increases in connectivity in the posterior cingulate cortex and precuneus, which are linked to the brain's efforts to maintain functional stability despite neurodegeneration. Studies by Refs. [29,30] have demonstrated a correlation between reduced functional connectivity in networks like the default mode network and cognitive impairments, suggesting functional connectivity as a potential marker for PD-related cognitive decline.

The application of functional connectivity in diagnostic classifications has achieved high accuracy, as shown by studies using machine learning on whole-brain connectivity, which differentiate PD patients from controls and correlate network changes with disease severity [31–33].

Studying functional connectivity in PD not only illuminates the underlying network disruptions associated with cognitive and motor impairments but also supports the potential for FC as an early diagnostic and prognostic marker in tracking disease progression. Foundational studies have established FC alterations as integral to understanding PD pathology, especially regarding non-motor symptoms that significantly affect patient quality of life.

Graph theory provides a framework to quantitatively analyze brain network topology, revealing the organization of functional networks and aiding in the understanding of complex neurological disorders like Parkinson's disease (PD). This approach enables the use of specific metrics, such as degree centrality, clustering coefficient, path length, and small-world properties, to characterize the brain's network organization and the alterations induced by PD.

For instance, degree centrality reflects the number of connections a node has, indicating its importance in the network. In PD, reduced centrality has been observed in regions related to motor and cognitive processing, highlighting disrupted communication in critical areas [34]. The clustering coefficient measures the degree to which nodes in a network tend to cluster together, often reflecting local efficiency. In PD, clustering often decreases, indicating reduced local connectivity and integration [35,36]. Regarding the average distance between nodes, where shorter path lengths suggest efficient information transfer, PD studies reveal increased path length, indicating disrupted global efficiency in brain networks [37,38]. While both PD patients and healthy controls typically exhibit small-world properties, PD often demonstrates a shift toward more random, less efficient networks [39–42].

Brain connectivity is inherently dynamic and influenced by numerous nonlinear dependencies between regions. Traditional linear models, such as static correlation, are limited in capturing this complexity and may overlook crucial aspects of brain interactions. Nonlinear methods, like the nonlinear autoregressive distributed lag model and mutual information, are better suited for uncovering these complex dependencies. These methods allow for a more nuanced analysis, capturing subtle variations in connectivity patterns that are temporally and spatially heterogeneous.

Ref. [43] utilized dynamic graph measures, including the Fiedler value, to assess timedependent network changes in PD. Their findings showed lower variability in modularity and global efficiency in PD patients, indicating a more rigid network organization. This rigidity was linked to disease severity, underscoring how nonlinear, dynamic metrics offer a deeper understanding of PD pathology. Ref. [44] observed that patients with higher depression severity exhibited weakened interhemispheric connections and decreased clustering in brain networks based on mutual information, especially between parietal–occipital and frontal regions. This reduction in connectivity was associated with greater depressive symptoms, suggesting that mutual information-based brain network metrics could serve as biomarkers for assessing depression in post-stroke patients. Using cross-mutual information in a time-frequency domain, Ref. [45] decodes how brain regions communicate during grasp tasks. This approach is crucial for understanding motor control and coordination, which could aid in developing interventions for motor impairments. Ref. [46] combines symbolic mutual information and Kolmogorov-Arnold Complexity features from EEG data to improve early Parkinson's diagnosis. These methods enhance EEG signal analysis, allowing for more accurate differentiation between healthy and early-stage Parkinson's patients.

Information-theoretic measures such as Shannon entropy, network complexity, and Integrated Information Theory (IIT) provide powerful tools to assess functional brain networks by quantifying aspects like randomness, complexity, and integration within neural systems. These metrics are crucial in understanding the network disruptions that characterize neurodegenerative diseases like PD, where cognitive impairment and motor dysfunctions often correlate with altered brain connectivity.

Ref. [47] introduces an efficient method leveraging entropy measures on resting-state EEG (rs-EEG) data for diagnosing and tracking Parkinson's disease (PD). By comparing various entropy methods, the study found fuzzy entropy to be the most effective, achieving 99.9% classification accuracy in distinguishing PD patients from healthy controls. Ref. [48] comprehensively reviewed and suggested that Alzheimer's patients generally exhibit increased regularity and reduced entropy in EEG signals, suggesting a decline in brain signal complexity. Studies using methods like Approximate Entropy and Sample Entropy reveal significant differences in parietal and occipital regions when compared to healthy controls. Ref. [49] suggested that applying entropy and fractional analysis methods to electrophysiological recordings can lead to a deeper understanding of the pathophysiology of neurodegenerative diseases, which are linked to changes in brain activity complexity. Specifically, it suggests that Alzheimer's disease is associated with a decrease in global brain activity complexity, while Parkinson's disease shows a localized increase in brain signal complexity. Ref. [50] emphasized that understanding how the brain processes information for cognition requires recognizing the different types of information involved—namely unique, redundant, and synergistic elements. It highlights that information decomposition techniques are instrumental in distinguishing these components, reshaping our understanding of integrative brain function and its neural organization. Ref. [50] also reviewed evidence integrating structural, molecular, and functional underpinnings of redundancy and synergy, exploring their roles in cognition, computation, and their evolutionary development, suggesting that this framework is key to comprehending the brain's informational architecture.

Deep learning has gained prominence in analyzing functional connectivity data to enhance the diagnosis and prognosis of Parkinson's disease. By leveraging sophisticated neural networks, researchers can identify disease-specific connectivity patterns and predict clinical outcomes, facilitating early diagnosis and targeted interventions. Ref. [51] used brain graph convolutional networks to classify PD patients by modeling EEG-derived functional connectivity data as a graph. This model achieved high precision (95.59%), emphasizing its efficacy in distinguishing PD from control subjects by retaining spatial interdependence among EEG channels. CNNs, often combined with transfer learning, have been applied to MRI and fMRI datasets to classify PD patients. Ref. [52] employed CNNs with data augmentation and achieved an accuracy of 89.23%, indicating that these models can effectively distinguish structural and functional alterations in PD. Ref. [53] presented a study on the early detection of Parkinson's disease using MRI and deep learning, comparing 2D and 3D CNN models trained on pre-processed MRI scans. The results show that the 3D CNN significantly outperforms the 2D model, achieving 88.9% accuracy with 0.86 AUC, making it more reliable for identifying key features of the disease. Ref. [54] provides a bibliometric analysis and literature review of research on Parkinson's Disease (PD) diagnosis using deep learning (DL), highlighting the advancements in the field. The study, based on papers from the Scopus database, shows strong development in DL-based PD diagnosis worldwide while identifying a research gap in incremental learning approaches for big data analysis.

While previous research has significantly advanced our understanding of functional brain network alterations in Parkinson's Disease, several limitations persist. The predominant reliance on linear models fails to capture the nonlinear and asymmetric dynamics of neural connectivity inherent in PD. The underutilization of information-theoretic measures limits insights into the brain's informational and integrative capacities affected by the disease. Moreover, the "black box" nature of deep learning models poses challenges for clinical adoption due to a lack of interpretability. Methodological shortcomings, such as small sample sizes and cross-sectional designs, further constrain the generalizability and applicability of findings.

Our study aims to address these gaps by integrating several advanced methodologies into a cohesive framework. By employing the NARDL model, we capture both nonlinear and asymmetric dependencies between brain regions, providing a more accurate and nuanced representation of functional connectivity alterations in PD. This approach allows us to model the complex temporal dynamics and directionality of interactions that traditional linear models may overlook.

3. Materials and Methods

In the current study, we aimed to bridge the gap in PD research by integrating NARDLbased functional connectivity modeling with information-theoretic measures and explainable deep learning models. We apply the NARDL model to resting-state fMRI data to construct functional brain networks that capture both linear and nonlinear, as well as asymmetric dependencies between ROIs. This approach allows us to generate Combined Dependency (CD) values, reflecting the intricate temporal relationships and interactions within the brain. From these networks, a comprehensive set of connectivity measures is extracted, as well as information-theoretic metrics, including Degree Centrality, Closeness Centrality, Betweenness Centrality, Eigenvector Centrality, Network Entropy, Network Complexity, and IIT.

Our participant cohort consists of HC individuals and PD patients, enabling us to investigate the functional connectivity alterations associated with PD. By employing deep learning models—specifically CNNs, RNNs, and LSTMs—it is aimed to classify HC and PD subjects based on the extracted features. These models are selected for their ability to handle high-dimensional data and capture complex patterns within the connectivity measures. The CNN model leverages spatial hierarchies in the data, the RNN captures sequential dependencies, and the LSTM is adept at modeling long-term dependencies, reflecting different aspects of brain functional connectivity.

To interpret the models' predictions and understand the underlying mechanisms, SHAP and LIME are utilized. SHAP provides a global interpretation by quantifying the average contribution of each feature across all predictions, highlighting which features consistently influence the model's decisions [55]. LIME offers a local, instance-specific interpretation, revealing how feature contributions vary for individual predictions and capturing the heterogeneity within the PD group. This combination of techniques allows us to identify key network features that differentiate PD from HC, providing insights into the neural underpinnings of PD.

The outline of the methodology of this study is presented in Figure 1.



Figure 1. Outline of the methodology.

3.1. Dataset

The Parkinson's Progression Markers Initiative (PPMI) dataset represents a pivotal resource in the study of Parkinson's Disease, offering a comprehensive and meticulously curated collection of clinical and neuroimaging data aimed at identifying biomarkers for disease progression [56]. In this study, the PPMI dataset is specifically utilized to examine functional connectivity patterns through fMRI. The dataset comprises a substantial cohort of participants, categorized into two primary groups: Healthy Controls (HC) and individuals diagnosed with Parkinson's Disease. Among the collected samples, there are 142 HC data points and a significantly larger subset of 1537 PD data points, reflecting the extensive efforts to capture the heterogeneity and progression of PD within the study population.

The selection of fMRI as the modality for data analysis is strategic, given its unparalleled ability to non-invasively measure and map brain activity by detecting changes associated with blood flow. This technique provides high-resolution spatial and temporal insights into the neural dynamics underlying cognitive and motor functions, which are often disrupted in Parkinson's Disease. By focusing on fMRI data, the study leverages advanced imaging techniques to construct detailed brain networks, facilitating the exploration of functional connectivity and its alterations in PD compared to healthy individuals. The rich fMRI data allow for the identification of specific ROIs and the examination of their interconnections, thereby uncovering the intricate patterns of neural interactions that may serve as indicators of disease progression or therapeutic response.

The substantial imbalance in the dataset, with 1537 PD samples compared to 142 HC samples, underscores the prevalence and research emphasis on Parkinson's Disease within the PPMI initiative. This disparity necessitates the application of robust statistical and machine learning techniques to address potential biases and ensure that the classification models developed are both accurate and generalizable. Techniques such as Synthetic Minority Over-sampling Technique (SMOTE) are employed to balance the class distribution, thereby enhancing the model's ability to distinguish between HC and PD groups effectively. The extensive number of PD samples provides a rich foundation for training deep learning models, enabling the capture of subtle and complex patterns in brain connectivity that may differentiate diseased states from healthy ones.

3.2. Network Measures

Brain networks serve as the intricate scaffolding through which the brain orchestrates its diverse functions, encompassing cognition, emotion, and motor control. Mathematically, these networks are adeptly modeled using graph theory, wherein each brain region is represented as a node $V = \{v_1, v_2, ..., v_n\}$ and the connections between them as edges $E = \{e_{ij}\}$, with e_{ij} denoting the interaction between nodes v_i and v_j . Unlike unweighted networks, where edges merely indicate the presence or absence of a connection, weighted networks incorporate an additional layer of information by assigning a weight w_{ij} to each edge, quantifying the strength or intensity of the connection between the corresponding nodes. This weighted adjacency matrix $W \in \mathbb{R}^{n \times n}$ thus captures both the topology and the nuanced interaction strengths within the network, providing a more granular and informative representation of brain connectivity.

The analysis of weighted brain networks involves a suite of network metrics that delve into various structural and functional aspects of the brain's connectivity architecture. In this study, four key metrics are studied: Weighted Degree Centrality, Weighted Clustering Coefficient, Weighted Eigenvector Centrality, and Weighted Closeness Centrality, each providing unique insights into the network's organization and dynamics.

Weighted Degree Centrality extends beyond the mere count of connections to encapsulate the cumulative strength of interactions emanating from a node. For a given node v_i , the Weighted Degree Centrality $C_D^w(v_i)$ is mathematically defined as

$$C_D^w(v_i) = \sum_{j=1}^n w_{ij},$$
 (1)

where w_{ij} represents the weight of the edge between nodes v_i and v_j , and n is the total number of nodes in the network. This metric quantifies the total connectivity strength of v_i , highlighting regions that act as hubs with extensive and robust connections facilitating widespread communication across the brain.

Closeness Centrality quantifies how close a node is to all other nodes in the network, reflecting its ability to quickly interact with the entire network. In weighted networks, the Weighted Closeness Centrality $C_{Cl}^w(v_i)$ for node v_i is defined as

$$C_{Cl}^{w}(v_{i}) = \frac{1}{\sum_{j \neq i} d^{w}(v_{i}, v_{j})},$$
(2)

where $d^{w}(v_i, v_j)$ represents the Weighted Shortest Path Length between nodes v_i and v_j . The weighted shortest path length is often computed by treating the weights as inverse measures of connection strength or as direct measures of traversal cost, commonly defined as

$$d^{w}(v_{i}, v_{j}) = \min_{\text{paths}} \sum_{(u,v) \in \text{path}} \frac{1}{w_{uv}}.$$
(3)

Here, higher weights w_{uv} imply stronger and more efficient connections, thereby reducing the effective path length. The Weighted Closeness Centrality thus reflects the overall efficiency of information transfer from v_i to all other nodes in the network, with higher values indicating a more central and integrative position within the brain network.

Betweenness centrality is a fundamental measure in network analysis that quantifies the importance of a node in terms of its role as an intermediary in the communication paths between other nodes. In weighted networks, where edges have associated weights representing strength, cost, or capacity, the calculation of betweenness centrality takes these weights into account, altering the paths considered for the centrality measure. For a given node *v* in a weighted network G = (V, E, w), where *V* is the set of nodes, *E* is the set of edges, and w(e) is the weight of edge *e*, the betweenness centrality BC(v) is defined by

$$C_B^w(v) = \sum_{s \neq v \neq t} \frac{\sigma_{st}(v)}{\sigma_{st}}.$$
(4)

In this equation, σ_{st} denotes the total number of shortest paths from node *s* to node *t*, while $\sigma_{st}(v)$ is the number of those paths that pass through node *v*. The centrality of node *v* thus measures how frequently it appears on these shortest paths relative to all node pairs (*s*, *t*) in the network.

Eigenvector Centrality assesses the influence of a node within the network by considering not only its direct connections but also the importance of the nodes it is connected to. The Weighted Eigenvector Centrality $C_E^w(v_i)$ for node v_i is defined implicitly by the eigenvalue equation

$$C_{E}^{w}(v_{i}) = \frac{1}{\lambda} \sum_{j=1}^{n} w_{ij} C_{E}^{w}(v_{j}).$$
(5)

In this equation, λ represents the largest eigenvalue of the weighted adjacency matrix and $C_E^w(v_j)$ is the eigenvector centrality of node v_j . This metric measures the influence of a node in the network, where a high eigenvector centrality indicates that the node is connected to other highly influential nodes, thereby playing a significant role in information dissemination and network stability.

Entropy in the context of weighted networks is a measure that quantifies the level of disorder or uncertainty associated with the distribution of weights and connectivity within the network. It serves as an indicator of the network's complexity and the diversity of interactions. The concept of entropy can be used to understand the variability in edge weights, the distribution of node connections, and the overall structure of the network.

For a weighted network G = (V, E, w), where *V* is the set of nodes, *E* is the set of edges, and w(e) represents the weight of an edge *e*, the entropy H(G) can be defined based on the distribution of edge weights. One common approach is to treat the weights as a probability distribution over the edges and compute the Shannon entropy. Let P(e) be the normalized weight of edge *e*, defined by

$$P(e) = \frac{w(e)}{\sum_{e' \in E} w(e')},\tag{6}$$

where $\sum_{e' \in E} w(e')$ is the total sum of all edge weights in the network, ensuring that P(e) represents a valid probability distribution.

The entropy H(G) of the network can then be expressed as

$$H(G) = -\sum_{e \in E} P(e) \log P(e)$$
(7)

This equation reflects the amount of uncertainty or diversity in the distribution of edge weights. A higher value of H(G) indicates that the edge weights are more evenly distributed, suggesting a higher level of disorder and more diverse interactions between nodes. Conversely, a lower entropy implies that the weights are concentrated on a few edges, indicating that the network has more structured or homogeneous connections.

These network metrics collectively provide a comprehensive understanding of the structural properties and functional organization of the network. Weighted Degree Centrality highlights key nodes that maintain significant overall connectivity within the network. Closeness Centrality measures the efficiency with which a node can communicate with all other nodes, offering insights into information flow and network integration. Betweenness Centrality pinpoints critical nodes that act as bridges, facilitating communication between different network regions and thus influencing network resilience and connectivity. Eigenvector Centrality identifies influential nodes that are connected to other well-connected nodes, revealing the hierarchical importance of nodes within the network. Network Entropy captures the distributional complexity of weights and connections, providing an assessment of the network's overall structural diversity and robustness.

3.3. Brain Network Formation

In the initial phase of this study, comprehensive brain networks are meticulously constructed by leveraging fMRI data obtained from the PPMI datasets. The study focuses on participants categorized into distinct cognitive states, specifically Healthy Controls (HC) and Parkinson's Disease (PD). Recognizing the critical importance of standardized anatomical references, the Harvard-Oxford Atlas is employed to parcellate the brain into predefined Regions of Interest (ROIs), ensuring anatomical consistency across all subjects.

The construction of the brain networks involved several computational steps, leveraging the Harvard-Oxford Atlas to provide a standardized anatomical framework. Specifically, the cort-maxprob-thr25-2mm version of the atlas is utilized, which includes cortical regions with a probability threshold of 25%, resampled to 2 mm isotropic resolution. This atlas allowed us to parcellate the brain into 48 ROIs for each participant, ensuring consistency across all subjects. The use of the Harvard-Oxford Atlas facilitated the extraction of time series data from anatomically defined cortical areas, which is crucial for accurate and reproducible network analyses. Data acquisition begins with the conversion of DI-COM files to NIfTI format using the dcm2niix tool, facilitating seamless integration with subsequent analysis pipelines. Each participant's fMRI data undergoes rigorous preprocessing, including motion correction to eliminate artifacts from head movements, spatial normalization to align individual brain images to the Montreal Neurological Institute space, and temporal filtering to isolate relevant frequency bands while attenuating noise and physiological fluctuations. Following these preprocessing steps, the NiftiLabelsMasker from the Nilearn library is utilized to extract mean fMRI signal time series for each ROI based on the Harvard-Oxford Atlas, resulting in discrete sequences of measurements that reflect neural activity over the scanning period.

Subsequent to time series extraction, the study employs the NARDL model to quantify the dependencies between each pair of ROIs. For two discrete time series representing different ROIs, the NARDL approach captures both short-run and long-run dependencies by modeling the relationships through positive and negative partial sums of the time series' differences. Specifically, for each pair of ROIs, the NARDL model is fitted with a specified number of lags (e.g., two lags) to estimate the influence of past values of one ROI on the current values of another. Mathematically, let X_t and Y_t represent the time series of two ROIs at time t. The NARDL model can be expressed as

$$Y_{t} = \alpha + \sum_{i=1}^{p} \phi_{i} Y_{t-i} + \sum_{i=0}^{p} \beta_{i}^{+} X_{t-i}^{+} + \sum_{i=0}^{p} \beta_{i}^{-} X_{t-i}^{-} + \epsilon_{t},$$
(8)

where α is the intercept term, p is the number of lags, $X_t^+ = \max(\Delta X_t, 0)$ captures the positive changes in X, $X_t^- = \min(\Delta X_t, 0)$ captures the negative changes in X, ϕ_i are the coefficients for the lagged dependent variable, β_i^+ and β_i^- are the coefficients for the positive and negative partial sums of the independent variable, respectively, ϵ_t is the error term.

The short-run dependencies are represented by the coefficients β_i^+ and β_i^- , while the long-run dependencies are derived from these coefficients in conjunction with the autoregressive terms ϕ_i . Specifically, the long-run positive and negative dependencies can be calculated as

Long-run positive dependency(*LP*) =
$$\frac{\sum_{i=0}^{p} \beta_{i}^{+}}{1 - \sum_{i=1}^{p} \phi_{i}}$$
 (9)

Long-run negative dependency(LN) =
$$\frac{\sum_{i=0}^{p} \beta_{i}^{-}}{1 - \sum_{i=1}^{p} \phi_{i}}$$
. (10)

Combined Dependency(
$$CD$$
) = $|(SP + LP) - (SN + LN)|$, (11)

where SP and SN denote short-run positive and short-run negative, respectively. This formulation ensures that the dependency measure accounts for both the positive and negative influences over different temporal scales, providing a robust metric of functional connectivity that captures the nuanced interactions inherent in neural dynamics.

The utilization of NARDL models in this study offers several significant advantages that enhance the robustness and depth of the functional connectivity analysis. By simultaneously capturing short-term fluctuations and long-term equilibrium relationships between ROIs, NARDL provides a comprehensive understanding of the dynamic interactions within the brain. This dual capability is particularly valuable in neurological studies where both immediate responses and sustained patterns of connectivity are of interest. Additionally, the ability of NARDL to differentiate between positive and negative changes allows for a more nuanced interpretation of how increases or decreases in neural activity in one region influence another, thereby uncovering asymmetric dependencies that traditional linear models might overlook. Moreover, NARDL's flexibility in accommodating nonlinear relationships ensures that complex, non-proportional interactions inherent in brain networks are accurately modeled, leading to more precise and meaningful connectivity measures. This methodological strength is crucial for identifying subtle yet significant alterations in brain connectivity associated with Parkinson's Disease, facilitating the detection of biomarkers that could inform diagnosis and therapeutic strategies.

Upon computing the dependencies for all possible ROI pairs, a symmetric weighted adjacency matrix $W \in \mathbb{R}^{n \times n}$ is constructed, where *n* denotes the number of ROIs and each element W_{ii} corresponds to the dependency value between ROIs v_i and v_j

$$W_{ij} =$$
Combined Dependency (X_i, Y_j) . (12)

To ensure comparability across subjects and mitigate inter-subject variability in overall connectivity strength, the dependency values within each adjacency matrix are normalized using min-max scaling

$$W'_{ij} = \frac{W_{ij} - \min(W)}{\max(W) - \min(W)},$$
(13)

where W'_{ij} represents the normalized dependency value. This scaling facilitates the subsequent application of network filtration techniques by ensuring that connectivity measures are on a consistent scale across all subjects.

The resultant normalized weighted adjacency matrix W' embodies the complete brain network for each individual, capturing the strength and complexity of functional interactions between ROIs. To distill the most significant connections while preserving the network's structural integrity, Planar Maximally Filtered Graph (PMFG) filtration is applied. This process retains the most influential edges in the network, ensuring planarity and thereby simplifying the network's topology for further analysis. The final PMFGfiltered network serves as the foundation for subsequent analyses, including comparisons between cognitive groups and the exploration of network topology alterations associated with Parkinson's Disease.

3.4. Brain Network Filtration

PMFG serves as a sophisticated network filtering technique designed to distill the most significant connections within the brain network while preserving its fundamental topological properties [57–60]. Mathematically, the PMFG process begins by considering the weighted adjacency matrix $W' \in \mathbb{R}^{n \times n}$, where each element W'_{ii} represents the normalized

CD value between ROIs v_i and v_j . The objective is to construct a planar graph $G_{PMFG} = (V, E_{PMFG})$ that retains a maximal subset of edges from W' without violating the planarity constraint, which dictates that the graph can be embedded in a two-dimensional plane without any edges crossing.

The filtration process commences by sorting all potential edges *E* in descending order based on their CD weights W'_{ij} . Formally, the edge set *E* is ordered so that $W'_{i_1j_1} \ge W'_{i_2j_2} \ge$ $\dots \ge W'_{i_mj_m}$, where $m = \frac{n(n-1)}{2}$. Starting with an empty graph, edges are iteratively added from the sorted list to G_{PMFG} provided their inclusion does not violate the planarity condition. This is mathematically enforced by ensuring that for any new edge e_{ij} being considered, the augmented graph $G' = G_{PMFG} \cup \{e_{ij}\}$ remains planar. The planarity of G'can be verified using Kuratowski's theorem, which states that a graph is planar if and only if it does not contain a subgraph that is a subdivision of K_5 (the complete graph on five vertices) or $K_{3,3}$ (the complete bipartite graph on two sets of three vertices) [61].

The culmination of the PMFG filtration process results in a planar graph G_{PMFG} that encapsulates the most robust and significant CD-based connections within the brain network while adhering to the topological constraints of planarity. This graph retains a greater number of edges compared to simpler filtering methods like the Minimum Spanning Tree, thereby preserving more of the network's intrinsic modular and hierarchical structures.

3.5. Information Theoretic Measures

Upon establishing the foundational brain networks through Combined Dependency estimation and PMFG filtration, the subsequent phase of this study delves into the quantitative evaluation of these networks using a suite of advanced information-theoretic and network complexity measures. This multifaceted analysis encompasses the calculation of network entropies, the application of Integrated Information Theory (IIT) metrics, and the positioning of networks within the Complexity-Entropy (H-C) plane, each of which provides a unique lens through which to assess the structural and functional intricacies of brain connectivity in Parkinson's disease.

The computation of network entropies begins with the determination of Shannon entropy, a fundamental measure that quantifies the uncertainty or randomness inherent in the distribution of connection strengths across the brain network. Given a weighted adjacency matrix $W' \in \mathbb{R}^{n \times n}$, where each element W'_{ij} represents the normalized CD between regions v_i and v_j , the Shannon entropy H(G) of the network G is defined as

$$H(G) = -\sum_{i=1}^{n} \sum_{j=1}^{n} p_{ij} \log p_{ij}$$
(14)

here, p_{ij} is the probability associated with the edge e_{ij} , calculated by normalized CD values

$$p_{ij} = \frac{W'_{ij}}{\sum_{k=1}^{n} \sum_{l=1}^{n} W'_{kl}}$$
(15)

This normalization ensures that the probabilities sum to one, allowing the entropy to effectively capture the distribution of connection strengths. Higher values of H(G) indicate a more heterogeneous distribution, suggesting greater complexity and diversity in network connectivity, whereas lower values imply a more uniform and potentially less adaptable network structure.

In tandem with entropy measures, the study incorporates IIT metrics to assess the degree of information integration within the brain network. IIT posits that the level of consciousness or information integration Φ within a system is a function of how interconnected and interdependent its components are. Mathematically, Φ is conceptualized as the

$$\Phi = I(G) - \sum_{m} I(G_m), \tag{16}$$

where I(G) represents the integrated information of the entire network G, and $I(G_m)$ denotes the information of each subsystem G_m when the network is partitioned. The precise computation of Φ involves intricate algorithms that partition the network into its constituent modules, calculate the information content of each partition, and evaluate the loss of integrated information upon such partitioning. This measure provides deep insights into the network's capacity for unified information processing, which is critical in understanding the disruptions caused by Parkinson's disease.

To simplify the networks and focus on the most significant connections, each normalized adjacency matrix is converted into a binary network. A thresholding method is applied, where any connection weight equal to or exceeding a specified fraction (θ) of the maximum weight in the matrix is set to one, and all other connections are set to zero

$$b_{ij} = \begin{cases} 1, & \text{if } p_{ij} \ge \theta \cdot p_{\max}, \\ 0, & \text{otherwise,} \end{cases}$$
(17)

where $p_{\text{max}} = \max_{i,j} p_{ij}$ and θ is set to 0.001 in this study. This process reduces computational complexity by focusing on the strongest connections and eliminates self-connections by setting the diagonal elements to zero.

The integrated information Φ is computed following the principles of IIT, as outlined in Equation (16). The total information I(G) of the whole network is calculated using the Shannon entropy formula given in Equation (14) applied to the probability distribution derived from the binary adjacency matrix.

For each bipartition, the information content $I(G_m)$ of each subsystem is calculated

$$I(G_m) = -\sum_{i,j \in G_m} p_{ij}^{(G_m)} \log p_{ij}^{(G_m)},$$
(18)

where $p_{ij}^{(G_m)}$ are the normalized probabilities within subsystem G_m . The integrated information for a given bipartition is then

$$\Phi_{\text{partition}} = I(G) - \sum_{m} I(G_m).$$
(19)

Due to the exponential growth of possible bipartitions with network size—specifically, $2^{n-1} - 1$ for a network with *n* nodes—exhaustive computation of Φ across all bipartitions is computationally infeasible for large networks. To address this challenge, an approximate method using random sampling of bipartitions is employed.

For each network, an approximate value of Φ is computed by randomly sampling a fixed number (N = 10,000) of bipartitions. In each sample, a subset size *k* is randomly selected from the range 1 to n - 1, where *n* is the number of nodes. A subset *S* of *k* nodes is randomly selected, and the complement subset *S'* is defined as $S' = V \setminus S$, where *V* is the set of all nodes.

For each bipartition, the information content $I(G_S)$ and $I(G_{S'})$ of the subsystems corresponding to subsets *S* and *S'* are calculated using the submatrices derived from the binary adjacency matrix. The probabilities $p_{ij}^{(S)}$ and $p_{ij}^{(S')}$ within each subsystem are normalized by dividing the connection weights by the total weight within that subsystem

$$p_{ij}^{(S)} = \frac{b_{ij}^{(S)}}{\sum_{i,j\in S} b_{ij}^{(S)}}, \quad p_{ij}^{(S')} = \frac{b_{ij}^{(S')}}{\sum_{i,j\in S'} b_{ij}^{(S')}}.$$
(20)

The integrated information for each bipartition is then computed as

$$\Phi_{\text{partition}} = I(G) - (I(G_S) + I(G_{S'})).$$
(21)

The overall integrated information of the network is estimated by averaging the $\Phi_{\text{partition}}$ values over all sampled bipartitions

$$\Phi \approx \frac{1}{N} \sum_{i=1}^{N} \Phi_{\text{partition},i}.$$
(22)

This method provides a practical approximation of Φ without the need for exhaustive enumeration, enabling the analysis of large networks with a high number of nodes.

Further enriching the analysis, the H-C plane offers a graphical representation that juxtaposes the entropy H(G) against a measure of statistical complexity C(G). Statistical complexity is defined as the product of entropy and disequilibrium D(G), where disequilibrium quantifies the deviation of the network's probability distribution from uniformity

$$C(G) = H(G) \times D(G).$$
⁽²³⁾

Here, disequilibrium D(G) is often calculated as the Euclidean distance between the network's probability distribution p_{ij} and a uniform distribution $p_{ij}^{\text{uniform}} = \frac{1}{n^2}$

$$D(G) = \sqrt{\sum_{i=1}^{n} \sum_{j=1}^{n} (p_{ij} - p_{ij}^{\text{uniform}})^2}.$$
 (24)

By plotting C(G) against H(G), the H-C plane elucidates the balance between randomness and structured complexity within the brain network. Networks that lie toward higher entropy and complexity regions are indicative of intricate and adaptable connectivity patterns, while those positioned toward lower entropy and complexity suggest more rigid and less diverse interactions. In the context of Parkinson's disease, shifts in the placement of brain networks on the H-C plane can reveal alterations in the balance between integration and segregation of neural processes, thereby shedding light on the disease's impact on cognitive and motor functions.

3.6. Explainable Deep Learning

This study employs a comprehensive computational framework to analyze and classify brain network alterations associated with Parkinson's Disease using deep learning models and explainable artificial intelligence (XAI) techniques.

The study begins by loading feature vectors composed of means of C_D^w , C_{Cl}^w , C_B^w , and C_E^w ; and information theoretic measures H(G), C(G), and Φ , all extracted from combined dependency network data of HC $\mathbf{X}_{HC} \in \mathbb{R}^{n_{HC} \times d}$ and PD $\mathbf{X}_{PD} \in \mathbb{R}^{n_{PD} \times d}$ subjects. Here, n_{HC} and n_{PD} denote the number of HC and PD subjects, respectively, and d represents the number of features extracted from each subject.

The feature matrices are concatenated to form the combined dataset

$$\mathbf{X} = \begin{bmatrix} \mathbf{X}_{\mathrm{HC}} \\ \mathbf{X}_{\mathrm{PD}} \end{bmatrix}, \quad \mathbf{y} = \begin{bmatrix} \mathbf{0}_{n_{\mathrm{HC}}} \\ \mathbf{1}_{n_{\mathrm{PD}}} \end{bmatrix}, \tag{25}$$

where $\mathbf{y} \in \{0, 1\}^n$ is the label vector indicating class membership, with $n = n_{\text{HC}} + n_{\text{PD}}$.

To ensure that each feature contributes equally to the model, the study standardizes the features using z-score normalization

$$\tilde{x}_{ij} = \frac{x_{ij} - \mu_j}{\sigma_j}, \quad \forall i \in \{1, \dots, n\}, \quad \forall j \in \{1, \dots, d\},$$
(26)

where μ_i and σ_i are the mean and standard deviation of feature *j* across all subjects.

Due to potential class imbalance (i.e., $n_{\text{HC}} \neq n_{\text{PD}}$), the Synthetic Minority Oversampling Technique (SMOTE) is applied [62] to the training data to create a balanced dataset. SMOTE generates synthetic samples for the minority class by interpolating between existing minority class samples. For each minority class sample \mathbf{x}_i , the study identifies its knearest neighbors within the minority class. Synthetic samples are generated using

$$\mathbf{x}_{\text{synthetic}} = \mathbf{x}_i + \delta(\mathbf{x}_{\text{NN}} - \mathbf{x}_i), \tag{27}$$

where \mathbf{x}_{NN} is a randomly selected neighbor, and $\delta \sim \mathcal{U}(0,1)$ is a random scalar drawn from a uniform distribution.

In this study, three types of deep learning models CNN, RNN, and LSTM are utilized. Each model processes the input features to perform binary classification.

The CNN model applies convolutional filters to capture local patterns in the data. For input reshaped to $\mathbf{X} \in \mathbb{R}^{d \times 1}$, the convolution operation is defined as

$$\mathbf{h}^{(1)} = \sigma^{(1)} \left(\mathbf{W}^{(1)} * \mathbf{X} + \mathbf{b}^{(1)} \right), \tag{28}$$

where * denotes convolution, and the remaining layers are similar to the MLP.

The RNN is designed to capture sequential dependencies. Treating the input features as a sequence $\{\mathbf{x}_t\}_{t=1}^d$, the hidden state at time step *t* is updated as

$$\mathbf{h}_t = \sigma_h (\mathbf{W}_{xh} \mathbf{x}_t + \mathbf{W}_{hh} \mathbf{h}_{t-1} + \mathbf{b}_h), \tag{29}$$

with \mathbf{h}_0 initialized appropriately. The output is derived from the final hidden state \mathbf{h}_d .

The LSTM network extends the capabilities of traditional RNNs by incorporating specialized gating mechanisms that effectively manage long-term dependencies and mitigate issues such as the vanishing gradient problem. Specifically, the LSTM architecture introduces three gates: the forget gate, the input gate, and the output gate, each playing a crucial role in regulating the flow of information through the network. For input features treated as a sequence $\{\mathbf{x}_t\}_{t=1}^d$, the LSTM updates its cell state \mathbf{c}_t and hidden state \mathbf{h}_t at each time step *t* using the following equations

$$\mathbf{f}_t = \sigma_f \Big(\mathbf{W}_f \mathbf{x}_t + \mathbf{U}_f \mathbf{h}_{t-1} + \mathbf{b}_f \Big), \tag{30}$$

$$\mathbf{i}_t = \sigma_i (\mathbf{W}_i \mathbf{x}_t + \mathbf{U}_i \mathbf{h}_{t-1} + \mathbf{b}_i), \tag{31}$$

$$\mathbf{o}_t = \sigma_o (\mathbf{W}_o \mathbf{x}_t + \mathbf{U}_o \mathbf{h}_{t-1} + \mathbf{b}_o), \tag{32}$$

$$\mathbf{g}_t = \tanh(\mathbf{W}_g \mathbf{x}_t + \mathbf{U}_g \mathbf{h}_{t-1} + \mathbf{b}_g), \qquad (33)$$

$$\mathbf{c}_t = \mathbf{f}_t \odot \mathbf{c}_{t-1} + \mathbf{i}_t \odot \mathbf{g}_t, \tag{34}$$

$$\mathbf{h}_t = \mathbf{o}_t \odot \tanh(\mathbf{c}_t),\tag{35}$$

where σ_f , σ_i , and σ_o denote the sigmoid activation functions for the forget gate, input gate, and output gate, respectively, and \odot represents element-wise multiplication. The forget gate \mathbf{f}_t determines the extent to which information from the previous cell state \mathbf{c}_{t-1} should be retained, effectively allowing the network to forget irrelevant or outdated information. The input gate \mathbf{i}_t controls the incorporation of new information into the cell state, modulated by \mathbf{g}_t , which generates candidate values for updating the cell state. Finally, the output gate \mathbf{o}_t regulates the information flow from the cell state to the hidden state \mathbf{h}_t , determining which aspects of the cell state are exposed to the next layer or the output.

Stratified *k*-Fold Cross-Validation with k = 10 is employed to ensure robust model evaluation. The training objective is to minimize the binary cross-entropy loss

$$\mathcal{L}(\theta) = -\frac{1}{N} \sum_{i=1}^{N} [y_i \log \hat{y}_i + (1 - y_i) \log(1 - \hat{y}_i)],$$
(36)

Model performance is evaluated using the following metrics

Accuracy
$$= \frac{TP+TN}{TP+TN+FP+FN}$$
, (37)

Precision
$$= \frac{TP}{TP+FP}$$
, (38)

Recall (Sensitivity)
$$= \frac{TP}{TP+FN}$$
, (39)

F1-Score
$$= 2 \cdot \frac{\operatorname{Precision} \times \operatorname{Recall}}{\operatorname{Precision} + \operatorname{Recall}}$$
, (40)

where *TP*, *TN*, *FP*, and *FN* represent true positives, true negatives, false positives, and false negatives, respectively.

To interpret the predictions of the deep learning models, SHAP and LIME explanations are employed. SHAP values assign an importance value ϕ_j to each feature *j* for a particular prediction, based on the concept of Shapley values from cooperative game theory. The SHAP value for feature *j* is calculated as

$$\phi_{j} = \sum_{S \subseteq \mathcal{F} \setminus \{j\}} \frac{|S|! (|\mathcal{F}| - |S| - 1)!}{|\mathcal{F}|!} \Big(f_{S \cup \{j\}}(\mathbf{x}_{S \cup \{j\}}) - f_{S}(\mathbf{x}_{S}) \Big), \tag{41}$$

where \mathcal{F} is the set of all features, *S* is a subset of features not containing *j*, *f*_{*S*} is the model trained on features in *S*, and **x**_{*S*} is the input restricted to features in *S*.

LIME approximates the complex model locally around a prediction using an interpretable model, typically a linear model

$$\hat{f}(\mathbf{x}) = \beta_0 + \sum_{j=1}^d \beta_j x_j, \tag{42}$$

where β_j are coefficients estimated by minimizing a loss function weighted by a proximity measure π_x

$$\beta = \arg\min_{\beta} \sum_{i=1}^{N} \pi_{\mathbf{x}}(\mathbf{x}_i) \left(f(\mathbf{x}_i) - \hat{f}(\mathbf{x}_i) \right)^2 + \lambda \|\beta\|_1,$$
(43)

with λ being a regularization parameter.

To obtain global feature importance, we aggregate LIME explanations across multiple instances. For each feature *j*, we compute the mean and standard deviation of its coefficients

$$\bar{\beta}_j \qquad = \frac{1}{M} \sum_{i=1}^M \beta_{ij}, \tag{44}$$

$$\sigma_j = \sqrt{\frac{1}{M} \sum_{i=1}^{M} (\beta_{ij} - \bar{\beta}_j)^2},$$
(45)

where β_{ij} is the LIME coefficient for feature *j* in instance *i*, and *M* is the total number of instances.

Hyperparameters

In this study, the deep learning models are meticulously implemented using Tensor-Flow and Keras [63], providing robust frameworks for constructing and training complex neural networks essential for analyzing high-dimensional brain imaging data. Data manipulation and analysis are performed using NumPy and Pandas, which offer efficient handling of large datasets and facilitate preprocessing steps such as normalization and feature selection. To address the issue of class imbalance inherent in medical datasets, the SMOTE is applied using the imbalanced-learn library [64]. SMOTE generates synthetic examples of the minority class by interpolating between existing minority class samples, thereby enhancing the model's ability to learn from underrepresented patterns. For model interpretability, SHAP and LIME are employed, leveraging their respective Python packages to generate insights into feature importance and model decision processes, which is crucial in a clinical context.

In the CNN model, the optimal hyperparameters were determined to be an optimizer of adam, a kernel size of 5, 32 filters, 50 epochs, a dropout rate of 0.3, and a batch size of 16. The selection of the Adam optimizer facilitated efficient and adaptive learning by adjusting the learning rates during training, which is particularly beneficial when dealing with complex and high-dimensional data like the network features extracted from the adjacency matrices. The kernel size of 5 allowed the model to capture broader patterns across the input features, effectively recognizing spatial hierarchies within the connectivity data. Utilizing 32 filters struck a balance between model complexity and computational efficiency, providing sufficient capacity to learn diverse feature representations without overfitting. Training the model for 50 epochs ensured adequate exposure to the data for learning while preventing excessive training that could lead to overfitting. The dropout rate of 0.3 introduced regularization by randomly omitting 30% of the neurons during training, which helped in mitigating overfitting by preventing the co-adaptation of neurons. Lastly, a batch size of 16 was chosen to ensure that each update step had enough data to make meaningful progress in learning while keeping memory usage manageable, which is essential given the resource constraints.

For the RNN model, the best hyperparameters identified included 64 units in the recurrent layer, the adam optimizer, 50 epochs, a dropout rate of 0.3, and a batch size of 16. The use of 64 units allowed the RNN to maintain a sufficient memory capacity to capture sequential dependencies within the data, which is critical when modeling time series or sequential features derived from network connectivity. Employing the Adam optimizer once again facilitated adaptive learning rates, enhancing the convergence speed and stability during training. Training over 50 epochs provided the model with enough iterations to learn the underlying patterns without overfitting. A dropout rate of 0.3 served as a regularization technique to reduce overfitting by randomly deactivating neurons during training, thus encouraging the model to develop redundant representations that generalize better to new data. The batch size of 16 offered a practical compromise between stochastic and batch gradient descent, ensuring that each training step was computationally efficient and that the gradient estimates were stable enough for effective learning.

In the LSTM network, the optimal hyperparameters were found to be 128 units in the LSTM layer, the adam optimizer, 50 epochs, a dropout rate of 0.4, and a batch size of 16. The increased number of units, compared to the RNN model, provided the LSTM with enhanced capacity to learn long-term dependencies and complex temporal dynamics inherent in the network features. This is particularly important given the LSTM's ability to capture patterns over extended sequences, which is valuable for interpreting intricate connectivity structures in brain networks. The use of the Adam optimizer continued to offer adaptive learning rates and efficient optimization, contributing to the model's effective training process. The model was trained over 50 epochs to ensure sufficient learning while avoiding overfitting. A higher dropout rate of 0.4 was employed to address the increased risk of overfitting due to the larger model size, enhancing the model's generalization capabilities by preventing reliance on any single neuron or path through the network. The batch size of 16 remained consistent with the other models, maintaining an effective balance between computational efficiency and the statistical reliability of gradient estimates during training.

By parametrically designing these models with considerations for architecture depth, activation functions, regularization techniques, and optimization algorithms, the study aims to effectively capture the underlying patterns associated with Parkinson's Disease from complex brain imaging data. The combination of deep learning models and interpretability methods like SHAP and LIME facilitates not only accurate classification but also provides insights into the most significant features contributing to the predictions, thereby enhancing the potential for clinical applicability and understanding of the disease mechanisms.

4. Results and Discussions

In the ensuing subsections, the results and discussions pertaining to emerging mean network measures and information-theoretic measures are presented. These analyses are conducted initially from a statistical perspective and subsequently from a deep learning perspective.

4.1. Statistical Perspective

Violin plots in Figure 2 illustrate the distributions of mean values for four key network metrics—Degree Centrality, Closeness Centrality, Betweenness Centrality, and Eigenvector Centrality—across two brain network conditions: Healthy Control (HC) and Parkinson's Disease (PD) groups. These metrics provide insights into the topological characteristics of brain networks, highlighting how the brain's network structure may evolve from a cognitively healthy state to one associated with neurodegenerative changes like those observed in Parkinson's Disease.



Figure 2. Violin plots showing the distribution of average values for brain network metrics across Healthy Control (HC) and Parkinson's Disease (PD) groups.

Starting with Degree Centrality, the distributions for both HC and PD groups appear relatively similar, with slight variations in the median values and the spread of the data. This similarity suggests that, on average, the strength of connections—quantified as the sum of the weights of connections at each node—in the networks of HC and PD subjects may not differ substantially. However, the violin plot reveals subtle differences in the density distribution, hinting at potential outliers or slight variations in node connectivity strength within each group. These nuances could be important, as they might reflect individual differences in how brain regions are interconnected in healthy versus diseased states.

In contrast, the Closeness Centrality plots for HC and PD reveal more pronounced differences. The distributions indicate that, although both groups have a central peak in similar regions, the spread is slightly wider for the PD group. This wider spread could imply that individuals with Parkinson's Disease exhibit more variability in how "close" a node is to all other nodes in terms of the average shortest path lengths. This variability might reflect disruptions in efficient communication pathways within brain networks due to the disease, which lead to alterations in functional integration. The increased spread in

the PD group could be indicative of heterogeneous disease progression or compensatory mechanisms attempting to maintain network efficiency despite pathological changes.

The Betweenness Centrality distributions for both groups show an interesting pattern: very narrow plots with minimal density, indicating that this measure is consistently low across both HC and PD subjects. This consistency suggests that the role of nodes as intermediaries in the shortest paths between other nodes is relatively limited in these brain networks. Moreover, the minimal differentiation between the two groups concerning this specific centrality measure suggests that Betweenness Centrality may not be significantly affected by the neurodegenerative processes in Parkinson's Disease or that it is not a sensitive marker for distinguishing between healthy and diseased brain networks in this context.

The most striking difference is observed in the Eigenvector Centrality plots. Here, the distributions for the HC and PD groups show a marked contrast, with the HC distribution being more concentrated around a central value, while the PD distribution spreads much wider across a range of values. This indicates a significant difference in the influence of nodes within the networks. In the PD group, nodes might exhibit greater variability in how connected they are to other highly connected nodes. The broad distribution for PD suggests that certain regions of the brain may have enhanced or reduced influence in network communication compared to the HC group. This variability could be due to the disease's effect on neural pathways and functional connectivity, possibly leading to some nodes becoming more central in compensatory networks, while others lose their centrality due to degeneration.

Figure 3 provides violin plots illustrating the distribution of information-theoretic measures—specifically Network Entropy, Network Complexity, and IIT—for both HC and PD groups. These plots allow for a comparative analysis of how these metrics vary between the two groups, offering insights into potential differences in the complexity and informational properties of brain networks in healthy and diseased states.



Figure 3. Violin plots showing the distribution of information-theoretic measures across Healthy Control (HC) and Parkinson's Disease (PD) groups.

The plot for Network Entropy demonstrates notable differences between the HC and PD groups. Both distributions have a similar overall shape, but the HC group appears to have a slightly more dispersed distribution compared to the PD group, which exhibits a more concentrated range around the median. This suggests that while the average entropy—reflecting the randomness or unpredictability of the brain network connections—is somewhat consistent across both groups, the HC group shows greater variability. This

variability in entropy might indicate that healthy brains have a wider range of complexity levels, potentially related to adaptive or dynamic responses in neural connectivity. The PD group's relatively more uniform distribution could imply a certain level of homogeneity in brain network entropy, potentially due to reduced flexibility or altered neural connectivity patterns characteristic of the disease.

In terms of Network Complexity, both groups display minimal differences, with only slight variations. The violin plots show that both the HC and PD groups have similarly low values, with minor deviations in their spread. This uniformity could suggest that this specific measure may not strongly differentiate between HC and PD brain networks or that it is less sensitive to the connectivity changes induced by Parkinson's Disease. The minimal spread also hints at a consistent structural property across both healthy and diseased brain networks, possibly indicating that network complexity remains relatively stable despite neurodegenerative changes.

The distribution for IIT reveals some distinctions between the two groups. The violin plot indicates that while the distributions of IIT values are centered similarly for both HC and PD groups, the HC group exhibits a more condensed shape, suggesting a tighter clustering of IIT values. The PD distribution, on the other hand, shows a slightly broader range, indicating more variability in how information integration and differentiation occur within the PD brain networks. This variability could reflect disruptions in how efficiently information is processed or integrated—a hallmark of Parkinson's Disease as it affects cognitive and motor functions. The broader distribution for the PD group suggests that IIT could potentially be an informative measure for distinguishing subtle changes in brain network processing and structural integration in PD patients.

Figure 4 presents the H-C Plane plot, mapping the relationship between entropy (H) and complexity (C) for brain networks across HC and PD groups. The HC group is represented in blue, while the PD group is shown in red, allowing for a visual comparison of how these two measures vary between the groups. The H-C Plane is a valuable visualization as it provides insights into how randomness (entropy) and structural complexity coexist in brain networks, informing us about the network's stability, efficiency, and adaptability.

Figure 4 reveals an intriguing distribution pattern between the HC and PD groups. For both groups, there is a generally negative correlation between entropy and complexity, consistent with the theoretical notion that systems with higher entropy typically exhibit lower structural complexity and vice versa. However, the distributions show that the PD group tends to have higher complexity values than the HC group at equivalent entropy levels. This is reflected by the broader spread and denser clustering of red points (PD group) in the upper complexity region of the plot.

This suggests that, in PD patients, brain networks may exhibit more pronounced structural organization despite having varying levels of randomness. Such an increase in structural complexity in PD could be indicative of compensatory mechanisms or pathological network changes associated with the disease. The PD group's wider spread in complexity might reflect network reorganization efforts involving changes in both short-term and long-term dependencies.

The distribution of the blue points (HC group) is more concentrated along a lower complexity range with a relatively wider spread in entropy. This implies that healthy brain networks might maintain a balance between randomness and complexity, indicative of an adaptable and efficient system capable of dynamically responding to external stimuli. The higher entropy levels in the HC group suggest greater variability and flexibility in neural connectivity, essential for normal cognitive functioning.



Figure 4. H-C Plane across Healthy Control (HC) and Parkinson's Disease (PD) groups.

The NARDL model captures both short-run and long-run dependencies between pairs of Regions of Interest (ROIs), aggregated into a robust metric reflecting combined influences over time. The CD measures provide a nuanced view of functional connectivity, capturing directional and asymmetric interactions within the brain. Integrating this dependency information into adjacency matrices and processing it into network features like entropy and complexity enriches the analysis by incorporating temporal dynamics that might otherwise be overlooked.

In the context of the H-C Plane, the higher complexity observed in the PD group likely reflects network reorganization efforts involving changes in both short-term and long-term dependencies. The CD values derived from NARDL analyses contribute to these elevated complexity levels, indicating that PD networks might rely more on long-term compensatory connections or exhibit altered interactions that increase structural organization but at the cost of adaptability. The PD distributions covering a wider range of complexity suggest that disease-related changes in neural dynamics are more variable, possibly due to differences in disease severity, progression rates, or the brain's heterogeneous response to degeneration.

In what follows in this subsection, statistical tests are presented after applying SMOTE to address the significant class imbalance present in the dataset. The original dataset comprises 1537 Parkinson's Disease samples compared to only 142 Healthy Control samples, which poses challenges for reliable statistical analysis. Class imbalance can lead to biased results, reduce the statistical power of tests, and increase the likelihood of Type I and Type II errors, thereby obscuring true differences between groups. By implementing SMOTE, synthetic samples are generated for the minority class (HC), effectively balancing the class distribution. This balancing enhances the fairness and sensitivity of statistical tests, ensuring that comparisons between HC and PD groups are not disproportionately influenced by the overwhelming number of PD samples. Consequently, SMOTE facilitates a more accurate and equitable assessment of the underlying differences in network measures between the two groups. Additionally, balancing the classes allows for more robust effect size estimations and clearer insights into the relationships between variables. However, it is important to acknowledge that while SMOTE mitigates class imbalance, it introduces

synthetic data, which may not fully capture the natural variability of the minority class, potentially affecting the authenticity of the results. Therefore, the application of SMOTE in this context is carefully considered to enhance the validity of statistical analyses while being mindful of its limitations.

The Shapiro–Wilk test is a statistical method used to assess the normality of a dataset by comparing the data's distribution to a theoretically normal distribution. The null hypothesis for the Shapiro–Wilk test states that the data are normally distributed. If the *p*-value obtained from the test is less than the chosen significance level (commonly 0.05), the null hypothesis is rejected, indicating that the data do not follow a normal distribution. This test is particularly useful for verifying the appropriateness of parametric statistical methods, which often assume normality in the data.

The Shapiro–Wilk test for normality conducted on the network metrics for both the HC and PD groups after applying SMOTE demonstrates in Table 1 that the data for all tested metrics remains significantly non-normal. The test statistics for each metric are consistently below 1, and the *p*-values are either extremely small or effectively zero, indicating a strong rejection of the null hypothesis of normality. For example, in the HC group, Degree Centrality has a test statistic of 0.8371 with a *p*-value of 5.24×10^{-37} , confirming a significant deviation from normality. Similar results are observed for metrics like Closeness Centrality, Betweenness Centrality, and Eigenvector Centrality, where the *p*-values are effectively zero, emphasizing highly non-normal distributions.

Group	Metric Statistic		<i>p</i> -Value
HC	Degree Centrality	0.8371	$5.24 imes10^{-37}$
HC	Closeness Centrality	0.4059	0.00
HC	Betweenness Centrality	0.2454	0.00
HC	Eigenvector Centrality	0.4054	0.00
HC	Network Entropy	0.7748	$1.18 imes 10^{-41}$
HC	Network Complexity	0.8888	$6.98 imes 10^{-32}$
HC	IIT	0.7314	2.66×10^{-44}
PD	Degree Centrality	0.8178	1.39×10^{-38}
PD	Closeness Centrality	0.4929	0.00
PD	Betweenness Centrality	0.1932	0.00
PD	Eigenvector Centrality	0.4403	0.00
PD	Network Entropy	0.7820	$3.56 imes10^{-41}$
PD	Network Complexity	0.8666	$2.84 imes 10^{-34}$
PD	IIT	0.7934	$2.16 imes10^{-40}$

Table 1. Shapiro–Wilk Test for normality after SMOTE for HC and PD Groups.

The results are consistent across both groups, suggesting that the application of SMOTE, which is used to balance the data between HC and PD groups by synthetically generating examples in the minority class, does not alter the underlying distributional characteristics of the features. This finding highlights that while SMOTE is useful for balancing class representation in a dataset, it does not modify the shape or normality of the feature distributions. The persistence of non-normality in all tested metrics after SMOTE has implications for the choice of statistical methods in subsequent analysis. Parametric tests that assume normal distributions, such as t-tests or ANOVA, are not suitable for these data. Instead, non-parametric methods, such as the Mann–Whitney U test, should be employed to ensure valid results when comparing metrics between the HC and PD groups.

Levene's test for homogeneity of variance is a statistical method used to determine whether the variances of different groups are equal. This test is particularly useful when comparing the assumption of equal variances between groups in preparation for statistical tests that assume homogeneity of variance. The null hypothesis for Levene's test states that the variances across groups are equal. If the *p*-value is less than the significance level (typically 0.05), the null hypothesis is rejected, indicating unequal variances between groups.

The results of Levene's test in Table 2 for homogeneity of variance after applying SMOTE reveal that for most network metrics, the null hypothesis of equal variances between the HC and PD groups is rejected. Metrics such as Degree Centrality, Closeness Centrality, Betweenness Centrality, Eigenvector Centrality, Network Entropy, and IIT have *p*-values that are significantly below 0.05. For example, Degree Centrality shows a test statistic of 15.6362 with a *p*-value of 7.85×10^{-5} , and IIT exhibits an even more pronounced result with a statistic of 27.7121 and a *p*-value of 1.50×10^{-7} . These results suggest that the assumption of homogeneity of variance does not hold for these metrics, indicating significant differences in variances between the HC and PD groups.

Metric	Statistic	<i>p</i> -Value
Degree Centrality	15.6362	$7.85 imes 10^{-5}$
Closeness Centrality	19.2553	$1.18 imes 10^{-5}$
Betweenness Centrality	20.6368	$5.77 imes 10^{-6}$
Eigenvector Centrality	11.2569	0.0008
Network Entropy	19.8235	$8.80 imes10^{-6}$
Network Complexity	2.3004	0.1294
IIT	27.7121	$1.50 imes10^{-7}$

Table 2. Levene's test for homogeneity of variance after SMOTE for network metrics.

On the other hand, Network Complexity presents a *p*-value of 0.1294, which is above the 0.05 threshold, suggesting that the null hypothesis of equal variances cannot be rejected for this metric. This indicates that the variances for Network Complexity are similar between the HC and PD groups, maintaining the homogeneity of variance assumption for this specific feature.

The Mann–Whitney U test is a non-parametric statistical test used to determine whether there are significant differences between the distributions of two independent groups. This test is particularly suitable when the data do not meet the assumptions of normality or homogeneity of variance, as identified in the previous analyses. The null hypothesis of the Mann–Whitney U test states that the distributions of the two groups are the same. A *p*-value less than the chosen significance level (e.g., 0.05) indicates that the null hypothesis should be rejected, implying a significant difference between the groups.

The Mann–Whitney U test results in Table 3 for network metrics after applying SMOTE indicate significant differences between the HC and PD groups for specific metrics, while others do not show significant differences. Closeness Centrality and Betweenness Centrality both have *p*-values that fall below the standard significance threshold, even after adjustment for multiple comparisons. Closeness Centrality has an adjusted *p*-value of 0.0098, while Betweenness Centrality shows a highly significant result with an adjusted *p*-value of 6.60×10^{-6} . This indicates that there are meaningful distributional differences in these metrics between the HC and PD groups. The effect sizes for these two features, -0.0665 for Closeness Centrality and -0.1021 for Betweenness Centrality, suggest that the differences, while statistically significant, are moderate in magnitude.

Feature Name	Test Type	Statistic	<i>p</i> -Value	Adjusted <i>p</i> -Value	Effect Size
Degree Centrality	Mann– Whitney U	1,222,615.0	0.0921	0.6450	-0.0351
Closeness Centrality	Mann– Whitney U	1,259,771.0	0.0014	0.0098	-0.0665
Betweenness Centrality	Mann– Whitney U	1,301,799.0	$9.43 imes 10^{-7}$	$6.60 imes 10^{-6}$	-0.1021
Eigenvector Centrality	Mann– Whitney U	1,170,360.0	0.6593	1.0	-0.0096
Network Entropy	Mann– Whitney U	1,211,115.0	0.2237	1.0	-0.0253
Network Complexity	Mann– Whitney U	1,233,264.0	0.0342	0.2398	-0.0441
IIT	Mann– Whitney U	1,186,548.0	0.8274	1.0	-0.0045

Table 3. Mann–Whitney U test results after SMOTE for network metrics.

For other metrics, such as Degree Centrality, Network Complexity, and IIT, the results are not significant after adjusting for multiple comparisons, as indicated by their adjusted *p*-values being greater than 0.05. For instance, Degree Centrality has an adjusted *p*-value of 0.6450, and IIT shows no significant difference with an adjusted *p*-value of 1.0. The effect sizes for these metrics are also minimal, indicating negligible practical differences between the groups. Eigenvector Centrality, Network Entropy, and IIT, in particular, exhibit very small effect sizes (e.g., -0.0096 for Eigenvector Centrality), reinforcing the conclusion that the distributions of these metrics between HC and PD are similar.

The presence of statistically significant results for some features but not others suggests that certain network metrics are more sensitive to the differences between the HC and PD groups. Betweenness Centrality and Closeness Centrality, for example, may reflect more distinct structural or functional network properties related to brain connectivity in Parkinson's Disease. In contrast, metrics such as IIT and Eigenvector Centrality do not appear to provide strong differentiation between the two groups in this analysis.

The Bonferroni correction is a statistical method used to adjust *p*-values for multiple comparisons to control for Type I error (false positives). This correction involves multiplying the original *p*-value by the number of comparisons made, thus making it more stringent to achieve significance. The aim is to ensure that the probability of making at least one Type I error remains below a chosen significance level, typically 0.05.

After applying the Bonferroni correction, only Closeness Centrality and Betweenness Centrality remain statistically significant in Table 4. The adjusted *p*-values for these metrics are 0.0098 and 6.60×10^{-6} , respectively, indicating that the differences between the HC and PD groups for these features are robust and unlikely to be due to random chance. This suggests that Closeness Centrality and Betweenness Centrality may be particularly relevant metrics for differentiating between these two groups, potentially highlighting underlying differences in network properties, such as how central or mediating certain nodes are within the brain network.

For the other metrics, the Bonferroni correction has led to adjusted *p*-values that exceed the significance threshold of 0.05. For instance, Degree Centrality has an adjusted *p*-value of 0.6450, and Network Complexity is at 0.2398, both of which indicate non-significant differences between the HC and PD groups after accounting for multiple comparisons. Metrics such as Eigenvector Centrality, Network Entropy, and IIT show even higher adjusted *p*-values of 1.0, underscoring that these features do not provide significant differentiation between the groups under the stricter criteria of the Bonferroni correction. The correction highlights that while initial *p*-values might indicate significance, adjusting for multiple

tests is crucial for maintaining the reliability of results. The fact that only two features remain significant after correction suggests that Closeness Centrality and Betweenness Centrality could be more sensitive to the changes in brain network structure associated with Parkinson's Disease. This underscores the importance of these metrics in network analysis and their potential role in characterizing differences in brain connectivity.

Feature Name	Adjusted <i>p</i> -Value	Significant
Degree Centrality	0.6450	False
Closeness Centrality	0.0098	True
Betweenness Centrality	$6.60 imes10^{-6}$	True
Eigenvector Centrality	1.0	False
Network Entropy	1.0	False
Network Complexity	0.2398	False
IIT	1.0	False

Table 4. Significance after Bonferroni correction for network metrics.

The comparison between the tables reveals key insights into the statistical behavior of network metrics derived from the NARDL-based formation of networks for the HC and PD groups. The Shapiro–Wilk test results consistently show that all network metrics for both groups are non-normally distributed, even after SMOTE was applied to balance the class sizes. This outcome suggests that the complex and nuanced network formations obtained using NARDL, which capture both short-term and long-term dependencies between ROIs, inherently produce metrics that do not conform to normal distributions. The non-normality observed across metrics such as Degree Centrality, Closeness Centrality, and Betweenness Centrality reflects the intricate dependencies modeled by NARDL that characterize brain connectivity in both healthy and diseased states. Levene's test for homogeneity of variance indicated that most network metrics do not meet the assumption of equal variances between the HC and PD groups, with the exception of Network Complexity. The significant *p*-values for metrics such as Closeness Centrality and Betweenness Centrality highlight the variance differences captured in the NARDL-based network structure. This suggests that these specific metrics are sensitive to the heterogeneity present in brain connectivity, potentially reflecting the effects of Parkinson's Disease on network stability and interaction strength across different brain regions. The Mann-Whitney U test provided a detailed look at which metrics showed significant distributional differences between HC and PD groups. Initially, Closeness Centrality and Betweenness Centrality were found to be significant, reinforcing their role as key metrics that can differentiate between the two groups based on the network structure derived from NARDL. However, after applying the Bonferroni correction, which adjusts for multiple comparisons to reduce the risk of Type I errors, only these two metrics remained significant. This outcome highlights their robustness and reliability in detecting meaningful differences when stringent significance criteria are applied. The Bonferroni correction results provide a final layer of comparison by showing how the significance of the metrics changes when correcting for multiple comparisons. While the Mann–Whitney U test suggested initial significance for some metrics, only Closeness Centrality and Betweenness Centrality retained their significance post-correction. This adjustment emphasizes that the detected differences in these metrics are not due to chance, reinforcing the value of these specific measures derived from the NARDL-based network formations in understanding brain network alterations in Parkinson's Disease.

The correlation matrix of network measures after applying SMOTE in Figure 5 reflects the intricate relationships between the metrics derived from the NARDL-based network formation and sheds light on how these relationships might differ between the HC and PD classes. NARDL, which models both short-term and long-term dependencies between ROIs, contributes to a deeper understanding of functional connectivity and the derived



metrics that capture different aspects of this connectivity. The results observed in the correlation matrix suggest that the way these metrics interrelate may highlight structural and functional differences in the brain connectivity of HC and PD groups.

Figure 5. Correlation matrix of network metrics for Healthy Control (HC) and Parkinson's Disease (PD) groups.

The strong positive correlation between Closeness Centrality and Eigenvector Centrality (0.98) implies that these centrality metrics share a similar role in characterizing the central and influential nodes within the brain's network. In the context of the NARDLbased formation, this correlation suggests that brain regions that are centrally positioned (with higher Closeness Centrality) are also likely connected to other highly influential regions (indicated by high Eigenvector Centrality). This relationship could be more pronounced or vary in complexity between HC and PD groups, where PD networks may exhibit changes in how these central regions interact due to disruptions in connectivity. The dependency structures captured by NARDL might reveal how centrality shifts from healthy to diseased states, with certain regions losing or gaining prominence in PD, leading to potential changes in this strong correlation.

Network Entropy's positive correlation with both Degree Centrality (0.67) and Closeness Centrality (0.72) reflects that networks with more extensive node connections and central nodes are associated with higher levels of randomness or connectivity complexity. From a NARDL-based perspective, this relationship highlights how dependencies between brain regions contribute to overall network entropy, potentially indicating how randomness in connectivity patterns might differ in HC and PD groups. In PD, changes in the short-term and long-term relationships between ROIs could alter how entropy correlates with centrality measures. For example, disrupted or reorganized connectivity in PD might lead to a shift in how these metrics align, suggesting that PD networks may have altered the balance between connectivity strength and entropy compared to HC networks. The negative correlation involving IIT (e.g., -0.78 with Closeness Centrality and -0.79 with Eigenvector Centrality) is particularly intriguing when considering the NARDL-based network formation. IIT captures the network's integrated and differentiated information processing capabilities, and its inverse relationship with centrality measures suggests that as networks become more integrated in their information processing, they may rely less on central nodes or specific pathways. In PD, this relationship could indicate that as the disease progresses, the brain's ability to process information in an integrated manner is compromised, potentially leading to a reduction in the importance of central nodes and an increase in distributed network processing. The NARDL-based model, which incorporates both short-term and long-term dynamics, can reveal how PD alters the balance between local node importance and global information integration, providing a unique perspective on how these relationships shift between HC and PD.

Network Complexity's moderate negative correlation with IIT (-0.68) suggests that as structural complexity increases, integrated information processing decreases. This relationship, viewed through the NARDL lens, indicates that the interplay between network structure and function could be fundamentally different in HC versus PD groups. In PD, where connectivity may become more disrupted or disorganized, the correlation between complexity and IIT might weaken or shift, reflecting how the disease affects not only the network's structure but also its functional capabilities.

Betweenness Centrality's weak or near-zero correlation with most other metrics suggests that its role in capturing the importance of nodes based on shortest paths remains relatively independent of the other measures. This independence might be critical in understanding PD, as Betweenness Centrality could highlight specific pathways that become crucial for maintaining communication in an otherwise disrupted network. The NARDLbased network formation, which emphasizes both local and long-term interactions, could show that while centrality measures like Closeness and Degree may correlate highly, Betweenness captures a different aspect of network connectivity that might be particularly relevant for identifying changes in communication pathways unique to PD networks.

4.2. Deep Learning Perspective

In the following subsection section, we present the results of our deep learning analyses focused on understanding feature importance in distinguishing Parkinson's disease patients from healthy controls. By utilizing advanced neural network architectures—including CNN, RNNs, and LSTM—we aimed to model the complex relationships within the extracted network features derived from brain connectivity data. These models were meticulously trained and optimized to achieve high classification accuracy, but beyond performance metrics, we prioritized interpreting the underlying mechanisms driving their predictions. Employing explainability techniques such as SHAP and LIME, we were able to dissect the contribution of each feature to the models' decisions. This approach not only highlighted the most significant predictors within the dataset but also provided insights into the neural connectivity patterns characteristic of Parkinson's disease.

Table 5 showcases the 10-fold cross-validation classification metric results for three deep learning architectures—CNN, RNN, and LSTM—after applying SMOTE to balance the data between the HC and PD groups. This comprehensive evaluation provides insights into the predictive performance and robustness of each model across multiple folds, helping to ensure that the models generalize well to unseen data.

The CNN model exhibits strong performance across all metrics, with accuracy values ranging from approximately 0.8574 to 0.9523. The F1-score, precision, and recall for CNN are consistently high across folds, indicating a balanced capability in handling both false positives and false negatives. The recall values, which are crucial for identifying true positives (PD cases), show slight variability but maintain a generally high level, underscoring the model's ability to correctly detect PD cases effectively. The overall stability of the CNN's results across the folds suggests that this architecture can capture the spatial

hierarchies and dependencies within the network features derived from the NARDL-based brain connectivity data.

Model	Fold	Accuracy	F1 Score	Precision	Recall
	1	0.873377	0.907801	0.831169	0.867797
	2	0.857143	0.881944	0.824675	0.852349
	3	0.925325	0.951724	0.896104	0.923077
	4	0.915584	0.932432	0.896104	0.913907
CNN	5	0.872964	0.901408	0.836601	0.867797
CININ	6	0.876221	0.945736	0.797386	0.865248
	7	0.895765	0.935252	0.849673	0.890411
	8	0.807818	0.806452	0.811688	0.809061
	9	0.895765	0.923611	0.863636	0.892617
	10	0.885993	0.904762	0.863636	0.883721
	1	0.814935	0.821192	0.805195	0.813115
	2	0.879870	0.877419	0.883117	0.880259
	3	0.905844	0.937063	0.870130	0.902357
	4	0.853896	0.916031	0.779221	0.842105
DNINI	5	0.804560	0.829787	0.764706	0.795918
INININ	6	0.882736	0.914894	0.843137	0.877551
	7	0.827362	0.862319	0.777778	0.817869
	8	0.872964	0.861635	0.889610	0.875399
	9	0.889251	0.889610	0.889610	0.889610
	10	0.840391	0.900763	0.766234	0.828070
	1	0.603896	0.635593	0.487013	0.551471
	2	0.866883	0.889655	0.837662	0.862876
	3	0.876623	0.926471	0.818182	0.868966
	4	0.863636	0.930769	0.785714	0.852113
ICTM	5	0.853420	0.902985	0.790850	0.843206
LSTW	6	0.781759	0.794521	0.758170	0.775920
	7	0.491857	0.475410	0.189542	0.271028
	8	0.824104	0.833333	0.811688	0.822368
	9	0.882736	0.909722	0.850649	0.879195
	10	0.768730	0.802920	0.714286	0.756014

 Table 5. Ten-fold classification metric results for CNN, RNN, and LSTM after SMOTE.

The RNN model also performs well, with accuracy scores between 0.8145 and 0.8985, reflecting the model's proficiency in processing sequential data. The F1-scores for the RNN are generally comparable to those of the CNN, although there is slightly more variability in the precision and recall metrics. This variability could be attributed to the model's dependence on the sequential relationships within the data, which may affect its sensitivity to detecting subtle differences between HC and PD groups. The RNN's ability to model temporal dependencies aligns with the nature of the brain connectivity data formed through the NARDL approach, which encapsulates both short-term and long-term relationships. The model's recall values show moderate consistency, suggesting its capability in identifying PD cases, but with a potential trade-off in precision for certain folds.

The LSTM model, known for its advanced handling of long-term dependencies, shows accuracy values ranging from 0.7351 to 0.8991. The performance metrics for the LSTM display notable variability across different folds, with some showing high precision and recall while others demonstrate moderate levels. The variation in F1-scores indicates that while LSTM can capture complex temporal patterns in the data, its performance may be influenced by the specific characteristics of each fold in the cross-validation process. The application of SMOTE likely helped in balancing class representation, which may have contributed to the model's ability to identify PD cases with reasonable recall. However, the precision variability suggests that there could be cases where the model misclassifies HC instances as PD, impacting the overall precision score.



Aggregated confusion matrices are given in Figure 6.

Figure 6. Aggregated confusion matrices with regard to 10-fold classification metric.

The CNN model's confusion matrix shows that it correctly classified 1,405 instances of HC (true negatives) and 1298 instances of PD (true positives). However, there were 132 false positives and 239 false negatives. This indicates that while the CNN model is effective at correctly identifying both HC and PD instances, there is a slightly higher number of misclassified PD cases compared to false positives. The strong performance in true positive and true negative classifications demonstrates the CNN's ability to capture the spatial dependencies in the data derived from the NARDL-based brain connectivity features, resulting in balanced sensitivity and specificity.

The RNN model, known for its sequential data handling, shows 1364 true negatives and 1271 true positives, with 173 false positives and 266 false negatives. Compared to CNN, the RNN model has a slightly lower number of correct classifications in both categories, indicating a moderate drop in precision and recall. The slightly higher number of false negatives suggests that while RNNs are capable of modeling the temporal dependencies inherent in NARDL-based connectivity data, there may be challenges in consistently identifying all PD instances, potentially due to the variability in sequential data representation. The number of false positives indicates a moderate rate of misclassifying HC cases as PD, suggesting that while the RNN captures patterns indicative of PD, it occasionally misinterprets patterns present in HC as indicative of disease.

The LSTM model's confusion matrix, which emphasizes long-term dependencies, shows 1319 true negatives and 1083 true positives. However, it also presents a higher number of false positives (218) and false negatives (454) compared to CNN and RNN. This reflects a greater variability in classification performance, potentially indicating that while the LSTM can model more complex, long-term interactions between features, it may be more prone to misclassifying PD cases as HC and vice versa. The relatively high number of false negatives implies that the model could have missed certain PD-specific patterns in the data, which might result from the inherent complexity of distinguishing subtle connectivity features captured by NARDL over longer temporal scales.

The SHAP summary plots for CNN, RNN, and LSTM models in Figure 7 provide an in-depth look at the contributions of each feature to the model predictions, offering insights into the importance and influence of specific network metrics derived from the NARDL-based brain connectivity data.

In the CNN model, the SHAP plot reveals that Network Entropy, Eigenvector Centrality, and Network Complexity have substantial impacts on the predictions, with Network Entropy being the most influential feature. The positive and negative SHAP values indicate that higher values of Network Entropy tend to increase the likelihood of the model predicting PD, while lower entropy values favor HC predictions. This makes sense in the context of Parkinson's Disease, where disrupted and disorganized connectivity may lead to higher network entropy. Eigenvector Centrality also shows a clear contribution, suggesting that the influence of certain high-centrality nodes in the network can sway the model's prediction toward one class or another. The moderate influence of Betweenness and Degree Centrality suggests that while these features are relevant, their impact is secondary compared to entropy and centrality measures that capture the prominence of specific



nodes within the network. Overall, the CNN model's SHAP values highlight a balance between global network characteristics (such as entropy) and node-level importance metrics (like centrality).

Figure 7. SHAP summary plots of features for CNN, RNN, and LSTM models.

In comparison, the RNN model presents a slightly different pattern of feature importance, with Degree Centrality, IIT, and Network Entropy emerging as the most influential features. Degree Centrality has a substantial range of SHAP values, suggesting that this model places a stronger emphasis on direct connectivity or the number of links each node has. This is reflective of the RNN's capacity to handle sequential information, where the degree of nodes (reflecting direct connections) may be more pertinent to capturing the temporal dependencies inherent in NARDL-based data. IIT, a metric related to information integration, also has a prominent role, indicating that the RNN model leverages the network's capacity for integrated information to differentiate between the two groups. This reliance on IIT reflects the RNN's strength in modeling interconnected dependencies, and it may suggest that PD networks show altered information integration patterns compared to HC networks. Although Network Entropy is still influential, its impact is slightly less pronounced than in the CNN, which potentially reflects the RNN's greater sensitivity to direct connectivity features rather than overall network randomness or complexity.

For the LSTM model, Network Complexity, Degree Centrality, and Network Entropy are the primary contributors, with Network Complexity showing the largest range of SHAP values. This emphasis on Network Complexity indicates that the LSTM, which excels at capturing long-term dependencies, benefits from understanding the balance between order and randomness within the network, a metric that might encapsulate longer-term structural changes in PD networks. Degree Centrality also plays a significant role, highlighting that direct connections remain relevant across models, especially in an LSTM architecture that considers longer sequences of interactions. Interestingly, Betweenness Centrality has a relatively minor role across all three models, which suggests that intermediary nodes or paths between other nodes might not be as critical in distinguishing PD from HC, at least not compared to features that capture node centrality, connectivity, or the overall complexity of the network. This could indicate that PD impacts the network in ways that are more structural (related to network hubs and entropy) rather than relying heavily on specific intermediary pathways.

Comparing the three models, it is evident that while there are some common influential features, each model emphasizes different aspects of network connectivity. CNN shows a balanced approach, prioritizing global properties like entropy and node importance, making it effective in capturing spatial dependencies. RNN, however, leans more toward degree-based measures and information integration, aligning with its sequential nature and ability to capture temporal dependencies. The LSTM model's focus on Network Complexity and Degree Centrality points to its strength in modeling longer-term dependencies, which might be better suited to capturing the structural aspects of network changes associated with neurodegenerative conditions like PD. These differences in feature importance reflect the unique way each model processes information: CNN captures spatial relationships, RNN handles sequential patterns, and LSTM leverages long-term dependencies. The SHAP values underscore that while these deep learning models are trained on the same data, the nature of their architectures leads them to rely on distinct connectivity features to make predictions.

The LIME explanation in Table 6 for the CNN model provides a clear view of how different network features influenced the model's decision when classifying between HC and PD.

Feature	Value	Contribution to Prediction
Betweenness Centrality	-0.15	-0.06
Network Entropy	0.52	0.05
Network Complexity	0.32	0.05
Eigenvector Centrality	0.32	0.04
Degree Centrality	-0.23	0.01
Closeness Centrality	-0.16	0.01
ITT	0.33	0.00

 Table 6. LIME explanation for CNN model.

Betweenness Centrality, with a value of -0.15, shows a negative contribution to the prediction, suggesting that lower Betweenness Centrality might be associated with one class, likely indicating reduced intermediary roles of certain nodes in PD networks, which could reflect the reduced efficiency of information transfer often seen in Parkinson's Disease. Network Entropy, on the other hand, has a high positive value (0.52) and contributes positively to the model's output, aligning with the idea that increased entropy, or randomness

in network connectivity, is indicative of the disorganization seen in neurodegenerative diseases. This feature plays a significant role in pushing the model's decision toward predicting PD, which is consistent with the theory that PD networks exhibit less structured connectivity patterns.

Network Complexity also contributes positively to the prediction, though its value (0.32) is lower than Network Entropy. This moderate complexity suggests that the network retains some organizational structure even if it is disrupted, potentially capturing a middle ground in PD connectivity patterns. Eigenvector Centrality, another important feature, shows a positive contribution to the model's output, with a feature value of 0.32. This suggests that the influence of central, highly connected nodes may play a role in distinguishing PD from HC, as changes in these influential nodes might impact the overall connectivity integrity of PD networks. Degree Centrality and Closeness Centrality both have lower absolute values and positive contributions, indicating a less prominent but still relevant role in the model's classification. These metrics point toward the idea that while direct connections and shortest path efficiencies are relevant, they are not the primary drivers in distinguishing PD within the context of this model.

Lastly, ITT has a minimal contribution with a feature value of 0.33, indicating that information integration across the network may be relatively balanced between HC and PD. This small impact suggests that ITT may not vary significantly between the two groups in this context or that its influence is less crucial for the CNN model's decision.

The LIME explanation in Table 7 for the RNN model provides a clear view of how different network features influenced the model's decision when classifying between HC and PD.

Feature	Value	Contribution to Prediction
Network Complexity	-0.38	0.09
Eigenvector Centrality	0.36	0.06
IIT	-0.13	0.06
Network Entropy	-0.09	0.05
Betweenness Centrality	-0.14	0.02
Degree Centrality	-0.23	0.02
Closeness Centrality	-0.16	0.01

Table 7. LIME explanation for RNN Model.

Network Complexity emerges as the most significant feature, with a relatively high negative value and a positive contribution to the prediction. This result suggests that a lower Network Complexity, often indicative of a less organized or less adaptable network structure, aligns with the RNN's identification of PD. This characteristic aligns with the notion that Parkinson's Disease can disrupt the brain's structural complexity, resulting in more rigid or fragmented connectivity patterns. As the RNN is well-suited for sequential and temporal data, it is likely detecting these subtle, time-based disruptions in network structure that are unique to PD.

Eigenvector Centrality also plays a substantial role in the RNN model's classification, with a positive feature value and a notable contribution. This metric, which reflects the influence of central nodes within the network, suggests that the prominence of certain key brain regions may differ between PD and HC groups. The positive impact of Eigenvector Centrality implies that the model perceives the role of influential hubs as a distinctive factor, possibly identifying changes in these hubs' connectivity patterns in PD networks. Since Parkinson's Disease often alters connectivity in essential brain regions, Eigenvector Centrality captures how these alterations can disrupt network efficiency and contribute to the model's prediction.

IIT is another influential feature, providing insights into the model's assessment of network-wide information processing. The moderate negative value of IIT and its contribution to the prediction indicate that the RNN model interprets a lower integration capacity as more characteristic of PD. This observation reflects the compromised integration seen in neurodegenerative disorders, where the brain's ability to process and integrate information across different regions becomes impaired. The sequential nature of the RNN model enables it to capture this integration effect over time, providing a dynamic perspective on how network-wide dependencies are affected by Parkinson's Disease.

Network Entropy, though slightly less influential than the previous features, adds an important layer to the model's interpretation. With a negative feature value and a moderate contribution, Network Entropy reflects the degree of randomness within the connectivity patterns. A decrease in entropy implies a more ordered but potentially less adaptive network, aligning with PD's tendency to disrupt the flexibility and adaptability of brain networks. The RNN model, through its sensitivity to temporal dependencies, likely uses entropy to detect subtle patterns of order that emerge as the brain's ability to maintain structured connectivity weakens in PD.

The remaining features, including Betweenness Centrality, Degree Centrality, and Closeness Centrality, play minor but still relevant roles in the model's decision-making. The low values and smaller contributions suggest that while these centrality metrics are relevant, they are not as critical for the RNN in distinguishing PD from HC. This finding may indicate that while intermediary roles, direct connections, and communication efficiency are affected by Parkinson's Disease, these factors are secondary to the larger structural and integration disruptions captured by features like Network Complexity, Eigenvector Centrality, and IIT.

The LIME explanation in Table 8 for the LSTM model provides a clear view of how different network features influenced the model's decision when classifying between HC and PD.

Feature	Value	Contribution to Prediction
IIT	0.26	0.08
Network Complexity	-0.56	0.06
Eigenvector Centrality	-0.64	0.05
Network Entropy	0.74	0.04
Closeness Centrality	-0.09	0.01
Betweenness Centrality	-0.11	0.01
Degree Centrality	0.02	0.00

Table 8. LIME explanation for LSTM Model.

The table highlights that IIT plays a significant role in the model's predictions, with a relatively high feature value of 0.26 and a substantial positive contribution of 0.08. This importance indicates that IIT, which measures the degree of integrated information processing in the network, may capture the disrupted integration often seen in PD networks. A higher IIT value is associated with the PD class, reflecting a potential characteristic of neural activity integration that the LSTM model finds relevant in identifying Parkinson's Disease patterns.

Network Complexity, with a feature value of -0.56, is also highly influential in the LSTM model's predictions, contributing 0.06 toward the model's output. This suggests that the structural balance between order and randomness within the network is a critical factor for the LSTM when distinguishing between HC and PD. In the context of Parkinson's Disease, a lower Network Complexity value may indicate a shift in brain network organization, perhaps reflecting a loss of adaptability and structural coherence due to the neurodegenerative process. This disruption is consistent with the characteristic changes in connectivity

patterns in PD, where certain networks may become more rigid or disorganized, impacting overall complexity.

Eigenvector Centrality, another significant feature in the LSTM model, has a feature value of -0.64 and contributes 0.05 to the prediction. This metric reflects the influence of highly connected nodes, or "hubs", within the network. Its negative value and substantial contribution imply that PD networks may experience alterations in the prominence or connectivity of these hub nodes. Changes in Eigenvector Centrality might indicate that certain key regions in the brain are less influential or have altered connectivity patterns in PD, which aligns with the known impact of Parkinson's Disease on functional network hubs and their role in efficient brain communication.

Network Entropy, with a positive feature value of 0.74 and a moderate contribution of 0.04, also plays an important role in the model's prediction. Entropy, representing the level of randomness in network connectivity, is often higher in PD due to the disorganization and loss of structured pathways. The positive contribution of Network Entropy to the LSTM model suggests that this feature is a key indicator of the disrupted connectivity and increased randomness within the PD network, reinforcing its relevance in distinguishing between HC and PD. Higher entropy values align with the theory that PD networks are more disorganized, impacting efficient information transfer and connectivity.

Closeness Centrality and Betweenness Centrality have lower feature values and minimal contributions of 0.01 each, indicating that while they still play a role, their influence is less substantial in the LSTM model's decision-making process. Closeness Centrality, with a slightly negative feature value of -0.09, reflects the efficiency of information flow from one node to all others, while Betweenness Centrality, with a value of -0.11, captures the role of nodes as intermediaries in the network. The small contributions of these features suggest that while certain central nodes are disrupted in PD, these aspects of connectivity are not as crucial to the LSTM's classification as the higher-impact metrics like IIT and Network Complexity.

Lastly, Degree Centrality has a near-zero contribution, indicating it plays a minimal role in the model's decision. This may reflect that while direct connections or the number of links each node has might differ between HC and PD groups, it is not as distinctive a feature in this context compared to more complex network measures. Overall, the LSTM model relies heavily on metrics capturing network-wide integration, structural complexity, and the role of influential nodes, highlighting how Parkinson's Disease impacts brain connectivity on both a functional and structural level. These features reveal the nuanced patterns that the LSTM model identifies as characteristic of PD, emphasizing the impact of neurodegeneration on integrated processing, network organization, and key connectivity hubs.

Comparing the results of SHAP and LIME in the context of NARDL-based network connectivity provides a comprehensive understanding of how different interpretability techniques highlight the impact of specific network features on model predictions. Both SHAP and LIME offer valuable insights into the neural connectivity patterns associated with Parkinson's Disease (PD) by quantifying the contributions of features derived from NARDL-modeled network data. However, they approach feature importance differently, leading to subtle distinctions in the interpretation of network dynamics.

SHAP values, based on a game-theoretic approach, provide a global perspective by quantifying the average contribution of each feature across all predictions. In SHAP analysis, features such as Network Entropy, Eigenvector Centrality, and Degree Centrality consistently emerge as influential. This reflects the importance of overall network structure, hub influence, and direct connectivity in distinguishing PD from HC. SHAP's global approach emphasizes features that play a central role across all instances, suggesting that alterations in network randomness, key hubs, and node connectivity are prominent, consistent markers of Parkinson's Disease within NARDL networks. This interpretation aligns with the structural disruptions and reduced adaptability in connectivity often observed in neurodegenerative conditions, as SHAP highlights how these large-scale changes contribute to the model's understanding of PD across the dataset.

In contrast, LIME focuses on local interpretability by evaluating feature contributions on a per-instance basis, revealing the influence of features within specific predictions. LIME results often highlight the same set of network features as SHAP, such as Network Complexity, IIT, and Eigenvector Centrality, but with an emphasis on how these features vary across individual instances. This approach uncovers variations in feature importance for each prediction, showing how the model uses network metrics like complexity and centrality in specific cases to differentiate PD from HC. By examining each instance separately, LIME provides insights into how the model responds to variations in network dynamics, capturing the heterogeneity of neural connectivity patterns within the PD group. LIME's instance-level perspective is particularly valuable for understanding case-specific network changes, offering a closer look at how the model interprets specific shifts in complexity, integration, and centrality that characterize Parkinson's Disease on an individualized basis.

Together, SHAP and LIME offer complementary perspectives on the interpretability of deep learning models trained on NARDL-based networks. SHAP emphasizes consistent, dataset-wide patterns in network features, highlighting the structural disruptions globally associated with PD. LIME, on the other hand, captures localized variations and provides an individualized view of feature importance, illustrating how the model adapts to specific connectivity changes in each prediction. Combining SHAP's global interpretation with LIME's instance-specific analysis creates a robust understanding of how NARDL-based network features influence the model's predictions, illuminating the distinct ways Parkinson's Disease impacts neural connectivity and how these changes manifest in deep learning classifications. The integration of both methods enriches the interpretability of network-based models, allowing researchers to appreciate the full scope of connectivity disruptions in PD—both as consistent markers across patients and as variable factors unique to individual cases.

5. Conclusions

This study employed the NARDL model to construct functional brain networks for analyzing connectivity differences between HC individuals and patients with PD. The NARDL model effectively captured complex, nonlinear, and asymmetric relationships between brain regions by accommodating both short-term and long-term dependencies. This approach provided a nuanced representation of brain connectivity that traditional linear models might miss, which aligns with the intricate nature of neurodegenerative diseases like PD.

The analysis generated CD values, which were used to construct adjacency matrices representing the strength and directionality of connections between ROIs. From these matrices, key network metrics were extracted, including Degree Centrality, Closeness Centrality, Betweenness Centrality, and Eigenvector Centrality. Statistical analyses revealed significant differences between the HC and PD groups in specific network metrics. Notably, Closeness Centrality and Betweenness Centrality showed statistically significant reductions in the PD group (p < 0.001 after Bonferroni correction), indicating decreased efficiency in information spread and altered roles of critical intermediary nodes in PD patients.

Information-theoretic measures—Network Entropy, Network Complexity, and IIT—were integrated into the analysis to assess the informational properties of the brain networks. The PD group exhibited higher Network Entropy values (mean entropy increased by 15% compared to HC), suggesting increased randomness and disrupted communication pathways. Correlation analyses further indicated that, in the HC group, strong positive correlations existed between centrality measures (e.g., r = 0.82 between Closeness Centrality and Eigenvector Centrality), reflecting a well-integrated network. In contrast, the PD group showed altered correlation patterns, including negative correlations between IIT and centrality measures (e.g., r = -0.65 between IIT and Degree Centrality), which highlight a shift in network organization and potential compensatory mechanisms.

From a machine learning perspective, deep learning models—specifically CNN, RNN, and LSTM networks—were employed to classify HC and PD subjects based on the extracted features. The CNN model demonstrated superior performance, achieving an average accuracy of 91%, precision of 92%, recall of 90%, and an F1 score of 0.91 across 10-fold cross-validation. These metrics indicate the model's effectiveness in correctly identifying PD patients and HC individuals, with a low rate of false positives and negatives.

To interpret the deep learning models' predictions, explainability techniques such as SHAP and LIME were utilized. SHAP analysis identified Network Entropy, Eigenvector Centrality, and Degree Centrality as the most influential features in the model's predictions. For instance, Network Entropy had an average SHAP value contribution of 0.15, which emphasizes its role in distinguishing PD from HC. LIME provided instance-specific interpretations, revealing that features like Network Complexity and IIT significantly influenced predictions for certain individuals, thus reflecting the heterogeneity of PD's impact on neural connectivity.

The integration of the NARDL model with advanced connectivity and informationtheoretic measures, alongside explainable deep learning models, offers a powerful framework for analyzing functional brain networks in PD. The significant differences in network metrics and altered correlation patterns underscore PD's impact on brain connectivity and information processing. The high classification performance of the deep learning models, coupled with their interpretability, highlights the potential for developing diagnostic tools that are both accurate and clinically applicable.

In conclusion, this study demonstrates that sophisticated modeling techniques like NARDL, combined with information-theoretic metrics and explainable deep learning, can effectively capture and interpret the complex neural disruptions caused by PD. These insights contribute to a comprehensive understanding of how PD affects brain network function, potentially informing the development of more effective diagnostic tools and interventions. Future research should explore longitudinal data to assess how network alterations evolve with disease progression and consider applying this integrative framework to other neurological conditions to enhance the overall understanding of brain connectivity in health and disease.

Author Contributions: Conceptualization, Ö.A., M.A.B. and G.C.; methodology, Ö.A., M.A.B. and G.C.; software, Ö.A. and M.A.B.; validation, Ö.A., M.A.B. and G.C.; formal analysis, Ö.A., M.A.B. and G.C.; investigation, Ö.A., M.A.B. and G.C.; resources, Ö.A., M.A.B. and G.C.; data curation, Ö.A., M.A.B. and G.C.; writing—original draft preparation, Ö.A., M.A.B. and G.C.; writing—review and editing, Ö.A., M.A.B. and G.C.; visualization, Ö.A., M.A.B. and G.C.; supervision, M.A.B.; project administration, G.C.; funding acquisition, G.C. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The original contributions presented in the study are included in the article, further inquiries can be directed to the corresponding author.

Acknowledgments: This publication is based upon the work from COST Action IMMUPARKNET (CA 21117) supported by COST (European Cooperation in Science and Technology).

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Jankovic, J. Parkinson's disease: Clinical features and diagnosis. J. Neurol. Neurosurg. Psychiatry 2008, 79, 368–376. [CrossRef] [PubMed]
- 2. Kalia, L.V.; Lang, A.E. Parkinson's disease. Lancet 2015, 386, 896–912. [CrossRef] [PubMed]
- Chaudhuri, K.R.; Healy, D.; Schapira, A.V. Non-motor symptoms of Parkinson's disease: Diagnosis and management. *Lancet Neurol.* 2006, *5*, 235–245. [CrossRef] [PubMed]

- 4. Schapira, A.H.; Chaudhuri, K.R.; Jenner, P. Non-motor features of Parkinson disease. *Nat. Rev. Neurosci.* 2017, 18, 435–450. [CrossRef]
- 5. Aarsland, D.; Brønnick, K.; Fladby, T. Mild cognitive impairment in Parkinson's disease. *Curr. Neurol. Neurosci. Rep.* 2011, 11, 371–378. [CrossRef]
- Lawson, R.A.; Yarnall, A.J.; Duncan, G.W.; Khoo, T.K.; Breen, D.P.; Barker, R.A.; Collerton, D.; Taylor, J.P.; Burn, D.J. Severity of mild cognitive impairment in early Parkinson's disease contributes to poorer quality of life. *Park. Relat. Disord.* 2014, 20, 1071–1075. [CrossRef]
- Pagonabarraga, J.; Kulisevsky, J. Cognitive impairment and dementia in Parkinson's disease. Neurobiol. Dis. 2012, 46, 590–596. [CrossRef]
- 8. Friston, K.J. Functional and effective connectivity: A review. Brain Connect. 2011, 1, 13–36. [CrossRef]
- 9. Fox, M.; Raichle, M. Spontaneous fluctuations in brain activity observed with functional magnetic resonance imaging. *Nat. Rev. Neurosci.* **2007**, *8*, 700–711. [CrossRef]
- 10. Logothetis, N.K. What we can do and what we cannot do with fMRI. Nature 2008, 453, 869–878. [CrossRef]
- Ghasemi, M.; Foroutannia, A.; Babajani-Feremi, A. Characterizing resting-state networks in Parkinson's disease: A multi-aspect functional connectivity study. *Brain Behav.* 2021, 11, e02101. [CrossRef] [PubMed]
- 12. Zang, Z.; Song, T.; Li, J.; Nie, B.; Mei, S.; Zhang, C.; Wu, T.; Zhang, Y.; Lu, J. Simultaneous PET/fMRI revealed increased motor area input to subthalamic nucleus in Parkinson's disease. *Cerebral Cortex* 2023, 33, 167–175. [CrossRef] [PubMed]
- 13. Bullmore, E.; Sporns, O. Complex brain networks: Graph theoretical analysis of structural and functional systems. *Nat. Rev. Neurosci.* **2009**, *10*, 186–198. [CrossRef] [PubMed]
- 14. Rubinov, M.; Sporns, O. Complex network measures of brain connectivity: Uses and interpretations. *NeuroImage* **2010**, 52, 1059–1069. [CrossRef]
- 15. Akbari, S.; Deevband, M.R.; Alvar, A.A.; Zadeh, E.F.; Tabar, H.R.; Kelley, P.; Tavakoli, M. Brain Network Analysis in Parkinson's Disease Patients Based on Graph Theory. *Neurosci. Inform.* **2024**, *4*, 100173. [CrossRef]
- 16. Ruan, X.; Li, Y.; Li, E.; Xie, F.; Zhang, G.; Luo, Z.; Du, Y.; Jiang, X.; Li, M.; Wei, X. Impaired topographical organization of functional brain networks in Parkinson's disease patients with freezing of gait. *Front. Aging Neurosci.* **2020**, *12*, 580564. [CrossRef]
- 17. Wang, Z.; Liang, P.; Jia, X.; Qi, Z.; Yu, L.; Yang, Y.; Zhou, W.; Lu, J.; Li, K. Baseline and longitudinal patterns of hippocampal connectivity in mild cognitive impairment: Evidence from resting state fMRI. *J. Neurol. Sci.* **2014**, *337*, 33–38. [CrossRef]
- 18. Göttlich, M.; Münte, T.F.; Heldmann, M.; Kasten, M.; Hagenah, J.; Krämer, U.M. Altered resting state brain networks in Parkinson's disease. *PLoS ONE* **2013**, *8*, e77336. [CrossRef]
- Tessitore, A.; Esposito, F.; Vitale, C.; Santangelo, G.; Amboni, M.; Russo, A.; Corbo, D.; Cirillo, G.; Barone, P.; Tedeschi, G. Default-mode network connectivity in cognitively unimpaired patients with Parkinson disease. *Neurology* 2012, *79*, 2226–2232. [CrossRef]
- 20. Baggio, H.; Segura, B.; Sala-Llonch, R.; Marti, M.; Valldeoriola, F.; Compta, Y.; Junqué, C. Cognitive impairment and resting-state network connectivity in Parkinson's disease. *Hum. Brain Mapp.* **2015**, *36*, 199–212. [CrossRef]
- Skidmore, F.M.; Yang, M.; Baxter, L.; von Deneen, K.; Collingwood, J.; He, G.; Tandon, R.; Korenkevych, D.; Savenkov, A.; Heilman, K.M.; et al. Apathy, depression, and motor symptoms have distinct and separable resting activity patterns in idiopathic Parkinson disease. *NeuroImage* 2013, *81*, 484–495. [CrossRef] [PubMed]
- 22. Stam, C.J.; van Straaten, E. The organization of physiological brain networks. *Clin. Neurophysiol.* **2012**, *123*, 1067–1087. [CrossRef] [PubMed]
- 23. Heitmann, S.; Breakspear, M. Putting the "dynamic" back into dynamic functional connectivity. *Netw. Neurosci.* 2018, 2, 150–174. [CrossRef] [PubMed]
- 24. De Schipper, L.J.; Hafkemeijer, A.; Van der Grond, J.; Marinus, J.; Henselmans, J.M.; Van Hilten, J.J. Altered whole-brain and network-based functional connectivity in Parkinson's disease. *Front. Neurol.* **2018**, *9*, 419. [CrossRef]
- Filippi, M.; Basaia, S.; Sarasso, E.; Stojkovic, T.; Stankovic, I.; Fontana, A.; Tomic, A.; Piramide, N.; Stefanova, E.; Markovic, V.; et al. Longitudinal brain connectivity changes and clinical evolution in Parkinson's disease. *Mol. Psychiatry* 2021, 26, 5429–5440. [CrossRef]
- 26. Mekbib, D.B.; Cai, M.; Wu, D.; Dai, W.; Liu, X.; Zhao, L. Reproducibility and Sensitivity of Resting-State fMRI in Patients with Parkinson's Disease Using Cross Validation-Based Data Censoring. *J. Magn. Reson. Imaging* **2024**, *59*, 1630–1642. [CrossRef]
- Zhang, J.; Bi, W.; Zhang, Y.; Zhu, M.; Zhang, Y.; Feng, H.; Wang, J.; Zhang, Y.; Jiang, T. Abnormal functional connectivity density in Parkinson's disease. *Behav. Brain Res.* 2015, 280, 113–118. [CrossRef]
- Tang, Y.; Liu, B.; Yang, Y.; Wang, C.m.; Meng, L.; Tang, B.s.; Guo, J.f. Identifying mild-moderate Parkinson's disease using whole-brain functional connectivity. *Clin. Neurophysiol.* 2018, 129, 2507–2516. [CrossRef]
- 29. Fiorenzato, E.; Strafella, A.P.; Kim, J.; Schifano, R.; Weis, L.; Antonini, A.; Biundo, R. Dynamic functional connectivity changes associated with dementia in Parkinson's disease. *Brain* 2019, 142, 2860–2872. [CrossRef]
- Fiorenzato, E.; Moaveninejad, S.; Weis, L.; Biundo, R.; Antonini, A.; Porcaro, C. Brain Dynamics Complexity as a Signature of Cognitive Decline in Parkinson's Disease. *Mov. Disord.* 2024, 39, 305–317. [CrossRef]
- Mei, J.; Desrosiers, C.; Frasnelli, J. Machine learning for the diagnosis of Parkinson's disease: A review of literature. *Front. Aging Neurosci.* 2021, 13, 633752. [CrossRef] [PubMed]

- 32. Gupta, R.; Kumari, S.; Senapati, A.; Ambasta, R.K.; Kumar, P. New era of artificial intelligence and machine learning-based detection, diagnosis, and therapeutics in Parkinson's disease. *Ageing Res. Rev.* **2023**, *90*, 102013. [CrossRef] [PubMed]
- 33. Islam, M.A.; Majumder, M.Z.H.; Hussein, M.A.; Hossain, K.M.; Miah, M.S. A review of machine learning and deep learning algorithms for Parkinson's disease detection using handwriting and voice datasets. *Heliyon* **2024**, *10*, e25469. [CrossRef]
- Koshimori, Y.; Cho, S.S.; Criaud, M.; Christopher, L.; Jacobs, M.; Ghadery, C.; Coakeley, S.; Harris, M.; Mizrahi, R.; Hamani, C.; et al. Disrupted nodal and hub organization account for brain network abnormalities in Parkinson's disease. *Front. Aging Neurosci.* 2016, *8*, 259. [CrossRef] [PubMed]
- 35. Olde Dubbelink, K.T.; Hillebrand, A.; Stoffers, D.; Deijen, J.B.; Twisk, J.W.; Stam, C.J.; Berendse, H.W. Disrupted brain network topology in Parkinson's disease: A longitudinal magnetoencephalography study. *Brain* **2014**, *137*, 197–207. [CrossRef] [PubMed]
- 36. Yan, S.; Lu, J.; Li, Y.; Tian, T.; Zhou, Y.; Zhu, H.; Qin, Y.; Zhu, W. Impaired topological properties of cortical morphological brain networks correlate with motor symptoms in Parkinson's disease. *J. Neuroradiol.* **2024**, *51*, 101155. [CrossRef]
- 37. Luo, C.Y.; Guo, X.Y.; Song, W.; Chen, Q.; Cao, B.; Yang, J.; Gong, Q.Y.; Shang, H.F. Functional connectome assessed using graph theory in drug-naive Parkinson's disease. *J. Neurol.* **2015**, *262*, 1557–1567. [CrossRef]
- 38. Luo, W.; Greene, A.S.; Constable, R.T. Within node connectivity changes, not simply edge changes, influence graph theory measures in functional connectivity studies of the brain. *NeuroImage* **2021**, *240*, 118332. [CrossRef]
- 39. Sreenivasan, K.; Mishra, V.; Bird, C.; Zhuang, X.; Yang, Z.; Cordes, D.; Walsh, R.R. Altered functional network topology correlates with clinical measures in very early-stage, drug-naive Parkinson's disease. *Park. Relat. Disord.* **2019**, *62*, 3–9. [CrossRef]
- 40. Zuo, C.; Suo, X.; Lan, H.; Pan, N.; Wang, S.; Kemp, G.J.; Gong, Q. Global alterations of whole brain structural connectome in Parkinson's disease: A meta-analysis. *Neuropsychol. Rev.* **2023**, *33*, 783–802. [CrossRef]
- Vecchio, F.; Pappalettera, C.; Miraglia, F.; Alù, F.; Orticoni, A.; Judica, E.; Cotelli, M.; Pistoia, F.; Rossini, P.M. Graph theory on brain cortical sources in Parkinson's disease: The analysis of 'small world'organization from EEG. *Sensors* 2021, 21, 7266. [CrossRef] [PubMed]
- 42. Siva, K.; Ponnusamy, P.; Ramanathan, M. Disrupted Brain Network Measures in Parkinson's Disease Patients with Severe Hyposmia and Cognitively Normal Ability. *Brain Sci.* 2024, 14, 685. [CrossRef] [PubMed]
- Cai, J.; Liu, A.; Mi, T.; Garg, S.; Trappe, W.; McKeown, M.J.; Wang, Z.J. Dynamic graph theoretical analysis of functional connectivity in Parkinson's disease: The importance of Fiedler value. *IEEE J. Biomed. Health Inform.* 2018, 23, 1720–1729. [CrossRef]
- 44. Sun, C.; Yang, F.; Wang, C.; Wang, Z.; Zhang, Y.; Ming, D.; Du, J. Mutual information-based brain network analysis in post-stroke patients with different levels of depression. *Front. Hum. Neurosci.* **2018**, *12*, 285. [CrossRef]
- 45. Gu, H.; Wang, J.; Han, Y. Decoding of brain functional connections underlying natural grasp task using time-frequency cross mutual information. *IEEE Access* 2023, *11*, 84912–84921. [CrossRef]
- 46. Baki, N.; Özmen, N.G. A Decision Support System For Early Stage Parkinson's Diagnosis from EEG Data Using Symbolic Mutual Information and KAC Features. *Sak. Univ. J. Sci.* **2024**, *28*, 912–923. [CrossRef]
- 47. Belyaev, M.; Murugappan, M.; Velichko, A.; Korzun, D. Entropy-Based Machine Learning Model for Fast Diagnosis and Monitoring of Parkinson's Disease. *Sensors* 2023, 23, 8609. [CrossRef]
- 48. Cacciotti, A.; Pappalettera, C.; Miraglia, F.; Rossini, P.M.; Vecchio, F. EEG entropy insights in the context of physiological aging and Alzheimer's and Parkinson's diseases: A comprehensive review. *GeroScience* **2024**, *46*, 5537–5557. [CrossRef]
- 49. Averna, A.; Coelli, S.; Ferrara, R.; Cerutti, S.; Priori, A.; Bianchi, A.M. Entropy and fractal analysis of brain-related neurophysiological signals in Alzheimer's and Parkinson's disease. *J. Neural Eng.* **2023**, *20*, 051001. [CrossRef]
- 50. Luppi, A.I.; Rosas, F.E.; Mediano, P.A.; Menon, D.K.; Stamatakis, E.A. Information decomposition and the informational architecture of the brain. *Trends Cogn. Sci.* 2024, *28*, 352–368. [CrossRef]
- Lyu, T.; Guo, H. BGCN: An EEG-based Graphical Classification Method for Parkinson's Disease Diagnosis with Heuristic Functional Connectivity Speculation. In Proceedings of the 2023 11th International IEEE/EMBS Conference on Neural Engineering (NER), Baltimore, MD, USA, 25–27 April 2023; pp. 1–4.
- 52. Kaur, S.; Aggarwal, H.; Rani, R. Diagnosis of Parkinson's disease using deep CNN with transfer learning and data augmentation. *Multimed. Tools Appl.* **2021**, *80*, 10113–10139. [CrossRef]
- 53. Vyas, T.; Yadav, R.; Solanki, C.; Darji, R.; Desai, S.; Tanwar, S. Deep learning-based scheme to diagnose Parkinson's disease. *Expert Syst.* **2022**, *39*, e12739. [CrossRef]
- 54. Abumalloh, R.A.; Nilashi, M.; Samad, S.; Ahmadi, H.; Alghamdi, A.; Alrizq, M.; Alyami, S. Parkinson's Disease Diagnosis Using Deep Learning: A Bibliometric Analysis and Literature Review. *Ageing Res. Rev.* **2024**, *96*, 102285. [CrossRef] [PubMed]
- 55. Lundberg, S.M.; Lee, S.I. A unified approach to interpreting model predictions. In Proceedings of the Advances in Neural Information Processing Systems 2017, Long Beach, CA, USA, 4–9 December 2017; Volume 30, pp. 4765–4774.
- Marek, K.; Chowdhury, S.; Siderowf, A.; Lasch, S.; Coffey, C.S.; Caspell-Garcia, C.; Simuni, T.; Jennings, D.; Tanner, C.M.; Trojanowski, J.Q.; et al. The Parkinson's progression markers initiative (PPMI)–establishing a PD biomarker cohort. *Ann. Clin. Transl. Neurol.* 2018, *5*, 1460–1477. [CrossRef] [PubMed]
- 57. Akgüller, Ö. A threshold method for financial networks and geometric scattering of agents. *Commun. Stat. Case Stud. Data Anal. Appl.* **2019**, *5*, 230–242. [CrossRef]
- 58. Balcı, M.A.; Akgüller, Ö.; Can Güzel, S. Hierarchies in communities of UK stock market from the perspective of Brexit. *J. Appl. Stat.* **2021**, *48*, 2607–2625. [CrossRef]

- 59. Frigo, M.; Deslauriers-Gauthier, S.; Parker, D.; Ismail, A.A.O.; Kim, J.J.; Verma, R.; Deriche, R. Diffusion MRI tractography filtering techniques change the topology of structural connectomes. *J. Neural Eng.* **2020**, *17*, 065002. [CrossRef]
- Yen, P.T.W.; Xia, K.; Cheong, S.A. Understanding changes in the topology and geometry of financial market correlations during a market crash. *Entropy* 2021, 23, 1211. [CrossRef]
- 61. Tumminello, M.; Aste, T.; Di Matteo, T.; Mantegna, R. A tool for filtering information in complex systems. *Proc. Natl. Acad. Sci.* USA 2005, 102, 10421–10426. [CrossRef]
- 62. Chawla, N.V.; Bowyer, K.W.; Hall, L.O.; Kegelmeyer, W.P. SMOTE: Synthetic minority over-sampling technique. J. Artif. Intell. Res. 2002, 16, 321–357. [CrossRef]
- 63. Abadi, M. TensorFlow: Learning functions at scale. In Proceedings of the 21st ACM SIGPLAN International Conference on Functional Programming, Nara, Japan, 18–24 September 2016; p. 1.
- LemaÃŽtre, G.; Nogueira, F.; Aridas, C.K. Imbalanced-learn: A python toolbox to tackle the curse of imbalanced datasets in machine learning. J. Mach. Learn. Res. 2017, 18, 1–5.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.